## Andrews University Digital Commons @ Andrews University

#### Dissertations

Graduate Research

2002

### The Relationship Between Spiritual Belief, Life Attitude, and Mental Health Among Physical Fitness Participants in Northern Indiana

Sharon K. Sacks Andrews University

Follow this and additional works at: https://digitalcommons.andrews.edu/dissertations

Part of the Other Religion Commons, and the Psychology Commons

#### **Recommended Citation**

Sacks, Sharon K., "The Relationship Between Spiritual Belief, Life Attitude, and Mental Health Among Physical Fitness Participants in Northern Indiana" (2002). *Dissertations*. 675. https://digitalcommons.andrews.edu/dissertations/675

This Dissertation is brought to you for free and open access by the Graduate Research at Digital Commons @ Andrews University. It has been accepted for inclusion in Dissertations by an authorized administrator of Digital Commons @ Andrews University. For more information, please contact repository@andrews.edu.



Seek Knowledge. Affirm Faith. Change the World.

Thank you for your interest in the

## Andrews University Digital Library of Dissertations and Theses.

Please honor the copyright of this document by not duplicating or distributing additional copies in any form without the author's express written permission. Thanks for your cooperation.

#### **INFORMATION TO USERS**

This manuscript has been reproduced from the microfilm master. UMI films the text directly from the original or copy submitted. Thus, some thesis and dissertation copies are in typewriter face, while others may be from any type of computer printer.

The quality of this reproduction is dependent upon the quality of the copy submitted. Broken or indistinct print, colored or poor quality illustrations and photographs, print bleedthrough, substandard margins, and improper alignment can adversely affect reproduction.

In the unlikely event that the author did not send UMI a complete manuscript and there are missing pages, these will be noted. Also, if unauthorized copyright material had to be removed, a note will indicate the deletion.

Oversize materials (e.g., maps, drawings, charts) are reproduced by sectioning the original, beginning at the upper left-hand corner and continuing from left to right in equal sections with small overlaps.

Photographs included in the original manuscript have been reproduced xerographically in this copy. Higher quality  $6^{\circ} \times 9^{\circ}$  black and white photographic prints are available for any photographs or illustrations appearing in this copy for an additional charge. Contact UMI directly to order.

ProQuest Information and Learning 300 North Zeeb Road, Ann Arbor, MI 48106-1346 USA 800-521-0600

# UMI®

Andrews University

School of Education

#### THE RELATIONSHIP BETWEEN SPIRITUAL BELIEF, LIFE ATTITUDE, AND MENTAL HEALTH AMONG PHYSICAL FITNESS PARTICIPANTS IN NORTHERN INDIANA

A Dissertation Proposal Presented in Partial Fulfillment of the Requirements for the Degree Doctor of Philosophy

> by Sharon K. Sacks June 2002

UMI Number: 3058315

## UMI®

#### UMI Microform 3058315

Copyright 2002 by ProQuest Information and Learning Company. All rights reserved. This microform edition is protected against unauthorized copying under Title 17, United States Code.

> ProQuest Information and Learning Company 300 North Zeeb Road P.O. Box 1346 Ann Arbor, MI 48106-1346

#### THE RELATIONSHIP BETWEEN SPIRITUAL BELIEF, LIFE ATTITUDE, AND MENTAL HEALTH AMONG PHYSICAL FITNESS PARTICIPANTS IN NORTHERN INDIANA

A dissertation presented in partial fulfillment of the requirements for the degree Doctor of Philosophy

by

Sharon K. Sacks

APPROVAL BY THE COMMITTEE:

Chair Carbonell Л. Member: Jimmy Kijai Member: Dennis Waite

S.

Paul

External:

Dean, School of Education /Graham Karen R.

7-8-02 Date approved

Kitchen

#### ABSTRACT

#### THE RELATIONSHIP BETWEEN SPIRITUAL BELIEF, LIFE ATTITUDE, AND MENTAL HEALTH AMONG PHYSICAL FITNESS PARTICIPANTS IN NORTHERN INDIANA

by

Sharon Sacks

Chair: Nancy J. Carbonell

#### ABSTRACT OF GRADUATE STUDENT RESEARCH

Dissertation

Andrews University

#### School of Education

#### Title: THE RELATIONSHIP BETWEEN SPIRITUAL BELIEF, LIFE ATTITUDE AND MENTAL HEALTH AMONG PHYSICAL FITNESS PARTICIPANTS IN NORTHERN INDIANA

Name of researcher: Sharon Sacks Name and degree of faculty chair: Nancy J. Carbonell, Ph.D. Date completed:

#### Problem

The purpose of this study is to explore spiritual beliefs as defined by an interconnectedness with self and others, generated from a relationship with a higher power, and consider its impact on attitude toward life and mental health.

#### Method

This study employed the survey research method to collect data investigating the relationships between spiritual beliefs, attitude toward life, and mental health.

A battery of three instruments was selected for this study. The Royal Free Questionnaire on Beliefs and Experiences, developed by King, Speck, and Thomas (1994), was used for measuring spiritual beliefs. The Optimism and Pessimism Questionnaire provided a global perspective of optimism and pessimism on the participants. The Mental Health Inventory (MHI) assessed the psychological distress or psychological well-being of the participants' focusing on the frequency or intensity of a psychological symptom during the past month.

#### Results

The relationship between spiritual belief and overall psychological well-being and the relationship between spiritual belief and depression were significant. As the commitment to a particular spiritual belief system strengthened, depression significantly decreased. The relationship between spiritual beliefs and loss of behavioral/emotional control was significant. Participants with a stronger commitment to a particular spiritual belief system reflected lower levels of uncontrollable behavior or emotion. General positive affect, emotional ties, and psychological distress also demonstrated significant relationships with spiritual belief. No relationships between spiritual beliefs and anxiety or life satisfaction were found.

#### Conclusions

There appeared to be no significant correlation between the spiritual beliefs scores and attitude toward life scores.

There was some indication that spiritual beliefs affected mental health. Participants presented with strong general positive affect, suffered significantly less psychological distress in their lives, were less depressed, more in control of their emotions and behaviors, and better able to establish emotional ties with others.

Spiritual belief, attitude toward life and mental health, as a set, were significantly related to age. Specific predictors include anxiety, loss of emotional/behavioral control, and psychological distress.

Only psychological distress and psychological wellbeing differentiated the men and women participants. Dedicated to my two amazing sons, Spencer and Aidan

#### TABLE OF CONTENTS

LIST OF TABLES	v
ACKNOWLEDGMENTS	i
Chapter	
I. INTRODUCTION	1
Purpose of the Study	
Research Questions	
Research Hypotheses	7
Significance of the Study	7
Conceptual Framework	8
Definitions of the Terms	0
Delimitation	
Limitations	
Organization of the Study	
II. REVIEW OF THE LITERATURE	4
Spiritual Dimension	
Attitude Toward Life	5
Mental Health and Wellness	
	-
III. METHODOLOGY	5
Design of the Study	
Sample	
Instrumentation	
The Royal Free Ouestionnaire on	
Beliefs and Experiences	7
Spiritual and Philosophical Scales 3	8
Reliability and Validity	
The Optimism and Pessimism Questionnaire 4	
Reliability and Validity 4	
Mental Health Inventory	
Reliability Validity 4	
Mental Health Subscales	
Global Mental Health Scales 4	
Mental Health Index	
Procedure	
Null Hypotheses	
Data Analysis 4	

iii

		S	ummar	y.	•	• •	•	•	•	•	•	•	•	•	·	•	•	•	•	·	•	49
•	IV. P	RESEN	TATIO	N ANI	DA	NAL	YS	IS	01	FI	DAT	ΓA		•	•		•	•	•	•	•	50
		De	scrip scrip atist	tive	Re	sul	ts	01	£ S	Sui	cve	≥y			•							
			Testi: mmary	ng .		•••		•	•	•	•				•							
			Y, DI: ENDAT												•			-	•	•		85
		Pu Ov Me Fi:	mmary rpose erview thodol nding:	of t w of logy s and	the the d D:	St e L  isc	udy ite us:	y era sic	atu ons	ire	• • •			• • •	• • •			• • •	• • •			86 87 89 89
		Imj	nclus: plicat commer	tions	5		•	•		•		•	•		•	•		-			•	96
	Appe	ndix																				
		A. B.	CORRE INSTE																			.01 .05
	REFE:	RENCE	LIST				•	•	•	•		•	•	•	•	•	•	•	•	•	1	.23
	VITA				•													•			1	31

#### LIST OF TABLES

1.	Percentage of Participants by Age and Gender 51
2.	Demographics of the Participants
3.	Distribution of Religious and Spiritual Understandings of Life
4.	Frequency Distribution of Religious Affiliations of the Respondents
5.	Level of Spiritual Belief of Participants 56
6.	Variables that Play a Role in Belief 56
7.	Frequency Distribution of Questions Not Included in Total Spiritual Belief Scale
8.	Mean Scores, Standard Deviations, Scale Ranges for Spiritual Belief Scale and Mental Health Scale
9.	Inter-correlation Matrix Between the 6 Spiritual Belief Variables, 2 Attitude Toward Life Variables,and 8 Mental Health Variables 61
10.	Multiple Regression Analysis of Spiritual Beliefs on Optimism
11.	Multiple Regression Analysis of Spiritual Beliefs on Pessimism
12.	Multiple Regression Analysis of Spiritual Beliefs on Anxiety
13.	Multiple Regression Analysis of Spiritual Beliefs on Depression
14.	Multiple Regression Analysis of Spiritual Beliefs on Loss of Behavioral/Emotional Control 69

v

15.	Multiple Regression Analysis of Spiritual Beliefs on General Positive Affect		•	•	•	71
16.	Multiple Regression Analysis of Spiritual Beliefs on Emotional Ties	•	•	•	•	72
17.	Multiple Regression Analysis of Spiritual Beliefs on Life Satisfaction		•	•		73
18.	Multiple Regression Analysis of Spiritual Beliefs on Psychological Distress	•	•	•	•	75
19.	Multiple Regression Analysis of Spiritual Beliefs on Psychological Well-being	•				76
20.	Multiple Regression Analysis of Spiritual Beliefs, Attitude Toward Life, and Mental Health Among Age		•	•	•	78
21.	Mean Levels and Standard Deviations of Spiritual Beliefs and Attitude by Gender .			•	•	79
22.	Mean Levels and Standard Deviations of Mental Health Variables by Gender			•		80
23.	Canonical Correlation Function 1		•	•	•	82

It is a humbling and deeply gratifying task to reflect on the many individuals whose encouragement and support made this project possible. Among the many to whom I am indebted, I would like to recognize and thank them for their invaluable contributions.

My Chair, Dr. Carbonell, and my committee members: Dr. Kijai, Dr. Waite, and Dr. Kitchen, for sharing their wisdom and knowledge.

Dr. Futcher, for believing in me from the very beginning.

My extraordinary family and friends, for their patience and generosity.

My parents, for a life time of support.

My husband, Tim, for his unconditional love and support throughout the years.

My "fairy Godmother," Susi, whose inner strength is a source of inspiration.

My children, Spencer and Aidan, who are my pride and joy, my blessing, and the true treasures of my life.

My Lord and Savior, for without Him, none of this could have been possible.

Thank You!!

vii

#### CHAPTER I

#### INTRODUCTION

Religion and spirituality are important in the lives of most Americans. National polls show that 9 out of every 10 Americans believe in God and consider religion important in their lives (Elkins, 1999; Kroll, 1995; Kroll & Sheehan, 1989). This is evidenced by nearly 500,000 churches, temples, and mosques in the United States, with a presence in virtually every community (Gallup, 1993). Surveys have found that 42% of Americans attend religious worship services somewhat regularly, while 67% are members of some local religious body (Gallup, 1993, 1995; Shafranske, 1996). Eighty-five percent of Americans have reported that religion is "very" or "fairly" important in their lives, while 95% profess a belief in God (Gallup, 1993; Shafranske, 1996). Clearly, these figures suggest that many individuals are seeking a spiritual connection at some level. However, despite the strong interest in spirituality, relatively few studies are found in this area compared to the need for understanding the impact of spirituality on mental health

(Shafranske, 1996). This may be in part due to the claim, by health professionals, that spirituality is a difficult concept to operationalize. Many researchers have suggested that spirituality cannot be defined, described fully, or researched and, therefore, is not a valid area of study (Goodloe & Arreola, 1992; Hawks, 1994; Hood, Spilka, Hunsberger, & Gorsuch, 1996; Ingersoll, 1998). This topic has often been criticized for fueling more controversy than valid research, effective interventions, or strategies (Elkins, 1999; Hamilton & Jackson, 1998; M. King, personal communication, August 20, 2000). Researchers know, as well, that there is probably little money available to study a topic such as spirituality. Additionally, it has been reported by the U.S. Department of Health and Human Services (1990) that the federal government spends more than 75% of its health-care dollars on chronic diseases while spending less than one half of 1% on preventive care for these diseases (Witmer & Sweeney, 1992).

Research shows that some mental health professionals may not feel as strongly about spirituality as most Americans. A survey conducted by Bergin and Jensen (1990) revealed that psychologists were among the least religious of the major mental health provider groups. Only 33% of clinical psychologists described faith as the most important

influence in their lives, as compared with 72% of the general population (Shafranske, 1996). Some health professionals, however, do view spirituality as a relevant area of inquiry with their clients. Shafranske and Maloney (1990) surveyed over 400 members of the American Psychological Association about their approach toward religion and psychology and found that nearly all respondents assessed the spiritual background of their clients. They also found that 57% used religious language or concepts with patients; 36% recommended participation in religion; 32% recommended religious or spiritual books; 24% prayed privately for a client; and 7% prayed with a client.

Although Hawks (1994) suggests that this is an area of health that is too often overlooked, Shafranske and Maloney's study (1990) suggests that more and more health professionals today believe that spirituality plays a relevant role, whether positively or negatively, in a person's mental health. Studies have often focused on religious attendance or religious involvement as a predictor of spirituality (Ellison & Levin, 1998). Chandler, Miner-Holden, and Kolander (1992) suggest that this may not be the best measure, claiming that spirituality is often independent of religion. Possibly, one might look to spiritual beliefs as a more accurate predictor of

spirituality. Many individuals may not profess a recognized religious faith or belong to an organized religious group, yet still consider themselves spiritual beings possessing spiritual beliefs. Therefore, studies focusing on only those affiliated with religious organizations may be excluding entire groups of individuals.

Richards and Bergin (1997) posit that "it is possible to be religious without being spiritual and spiritual without being religious" (p. 13). Taking all of this into consideration, for purposes of this study, spirituality is interpreted as an interconnectedness with self and others, generated from a relationship with a higher power. This is not to be misinterpreted as religiosity, which, for purposes of definition, tends to be more denominational, external, cognitive, behavioral, ritualistic, and public (Richards & Bergin, 1997).

As mentioned earlier, there has recently been an increased awareness of the impact of spirituality on mental health. A growing body of research now supports the health benefits of spirituality, prompting several authors to aggressively argue the need for spiritual health education in a variety of settings, including: the worksite (Bensley, 1991; Chapman, 1986), the university (Reichle, Vantrease, & Fink, 1989; Richardson & Nolan, 1984), the school system

(Hoyman, 1962; Jose, 1987; Osman & Russell, 1979), the community (Fahlberg & Fahlberg, 1991; Goodloe & Arreola, 1992; Seaward, 1991), and the primary care setting (Aldridge, 1991; McKee & Chappel, 1992).

Along with the awareness of the role spirituality plays in mental health, there has also been growing evidence that attitude may influence one's mental health (Seligman, 1998). It is suggested that individuals with an optimistic attitude are more apt to experience better health, less depression, higher achievement, and develop better health habits (Seligman, 1998). Thinking more optimistically has also been linked with better academic and work performance (Seligman, 1998).

Although, it may appear that life attitude and spiritual belief share common benefits with mental health, little data have been gathered that looks at the effects of spiritual belief on life attitude. It is my opinion that one's spiritual belief has a significant impact on one's outlook on life, affecting one's optimistic or pessimistic view. This perception, in turn, may create an attitude toward life that could be associated with a particular mental health profile.

Studies on how one's spiritual beliefs contribute to one's life attitude and to one's mental health are, thus, needed.

#### Purpose of the Study

The literature shows that spirituality has been an often overlooked dimension of mental health (Hawks, 1994) and may actually be an important component of optimum wellness. The purpose of this study is to explore this dimension of spirituality and assess its effects on one's attitude toward life and mental health.

#### Research Questions

This study examined the following questions:

1. Do people with higher levels of spirituality have a better attitude toward life?

2. Do people with higher levels of spirituality report better mental health?

3. Is age related to spiritual belief, attitude toward life, and mental health?

4. Is there a difference in spiritual belief, attitude toward life, and mental health between men and women?

5. Is there a linear relationship between spiritual belief, attitude toward life, and mental health?

#### Research Hypotheses

These research questions led to the following research hypotheses.

Hypothesis 1. There is a significant relationship between spiritual belief and attitude toward life.

Hypothesis 2. There is a significant relationship between spiritual belief and mental health.

Hypothesis 3. There is a significant relationship between spiritual belief, attitude toward life, mental health, and age.

Hypothesis 4. There is a significant difference in spiritual belief, attitude toward life, and mental health between men and women.

Hypothesis 5. There is a significant linear relationship between spiritual belief, attitude toward life, and mental health.

#### Significance of the Study

Because there is growing evidence that spiritual health exerts a major influence on the emotional, social, intellectual, and physical well-being of all persons (Goodloe & Arreola, 1992), it is believed that this study will provide a better understanding of the type of impact spirituality has on mental health. Informing health

professionals of the influences that spiritual beliefs play on one's attitude toward life and on one's mental health may aid them in the development of more effective strategies for intervention and preventive measures toward optimum mental health.

#### Conceptual Framework

Although spirituality continues to lack clarity in definition and application, many researchers agree that it is an essential element to wellness (Chandler et al., 1992; Goodloe & Arreola, 1992; Witmer & Sweeney, 1992). Many models of wellness (Chandler et al., 1992; Crose, Nicholas, Gobble, & Frank, 1992; Goodloe & Arreola, 1992; Hettler, 1979; Witmer & Sweeney, 1992) suggest that the spiritual component is as important as any other dimension of health.

Witmer and Sweeney (1992) go a step further by claiming that spirituality is not only an important component, but lies at the very core of wellness. They define spirituality as a "representation of values that reflect what is considered sacred and essential for the sustenance of life" (Witmer & Sweeney, 1992, p. 141). They consider several components as contributors to the dimension of spirituality: a sense of wholeness, inner wisdom, higher consciousness, meaning in life, hope or optimism for the future, values for

guiding human relationships, and decision making. They postulate that moral values guide one's behavior for wellbeing while demonstrating respect and compassion for the good of others.

Additionally, one's attitude toward life also appears to contribute significantly toward achieving optimum health and functioning (Seligman, 1998; Goodloe & Arreola, 1992). Maintaining a positive attitude toward life can often be difficult, especially when individuals are trying to meet the everyday demands of their hectic lives. Therefore, an awareness of one's attitude toward life and how it might affect one's mental health and well-being appears to be important. Studies suggest that individuals who perceive their lives to be more meaningful, purposeful, hopeful, and optimistic tend to experience better health (Goodloe & Arreola, 1992; Peterson, 2000; Seligman, 1998; Witmer & Sweeney, 1992). Witmer, Rich, Barcikowski, and Maque (1983) identified optimism as one of the prime variables that characterized individuals who experienced less anxiety and fewer physical symptoms.

Further, results from an intervention program that taught grade-school children to be more optimistic suggested that subsequent episodes of depression were less likely (Peterson, 2000). Optimism has also been linked to positive

mood and good morale; to perseverence and effective problem solving; to academic, athletic, military, occupational, and political success; to popularity; and to long life (Peterson, 2000).

These studies, along with personal observation and experience, lead me to believe that an individual is more likely to achieve a fulfilling life if one's perspective toward life encompasses hope for the future, strong spiritual beliefs, and a strong sense of self. These perspectives formed the theoretical foundation for this study.

#### Definitions of the Terms

Spirituality: An interconnectedness with self and others generated from a relationship with a higher power.

Spiritual belief: A culmination of one's convictions and perceptions gained through one's spiritual and life experiences.

Spiritual commitment: Abiding to one's spiritual belief system through application in everyday life (You are living what you believe).

Spiritual experience: An occurrence that precedes a connection between an individual and a higher being generated from one's spiritual commitment, spiritual

beliefs, or a plethora of other spiritual happenings.

Religion: The actual practice of faith; observance of rituals and requirements (e.g., church attendance).

Positive attitude toward life: The perception that one will experience promising outcomes to events and circumstances.

Negative attitude toward life: The perception that one will experience bleak outcomes to events and circumstances.

Mental health: A state of mind reflecting one's overall level of psychological well-being and/or psychological distress.

View of life: The strength of one's commitment to a particular spiritual belief system.

Practice of belief: The importance of practicing one's belief system.

Day-to-day influences: The level of belief in a spiritual force that influences daily decisions.

Cope with events: The level of belief that a spiritual force enables one to cope with what life bestows.

World Affairs: The level of belief that a spiritual force influences world affairs, e.g., wars.

Natural disasters: The level of belief that a spiritual force influences natural disasters.

#### Delimitation

The sample was delimited to individuals who were current or past members of a particular fitness center located in Northern Indiana.

#### Limitations

Limitations of this study include: not knowing whether or not the members would respond to the survey, how members might respond to the survey questions, and whether or not respondents would understand the meaning of the terms used within the instruments.

#### Organization of the Study

This study contains five chapters, which are arranged in the following manner:

Chapter 1 introduces the research topic, provides a statement of the problem, indicates the purpose, significance, and theoretical framework of the study, as well as provides the research questions, delimitations, limitations, and an outline of the study.

Chapter 2 contains a review of the literature related to spirituality and its impact on life attitude and mental health. Chapter 3 outlines the methodology of the research, including a description of the population, sampling procedures, procedures used for data collection, the reliability and validity of the instrument, research questions, and statistical analysis.

Chapter 4 describes the demographic data, the analysis of the data, and the summary of the results of data analysis.

Chapter 5 provides a summary of the findings, conclusions, implications, and recommendations for future research.

#### CHAPTER II

#### REVIEW OF THE LITERATURE

Science without religion is lame, religion without science is blind. --Albert Einstein

The review of the literature is divided into three sections. The first section discusses the spiritual dimension and its impact on mental health. The second section discusses attitude toward life with an emphasis on optimism and pessimism. And, lastly, the third section discusses mental health and wellness.

#### Spiritual Dimension

Today, national polls show that 9 out of every 10 Americans believe in God and consider religion either "very" or "fairly important" in their lives (Elkins, 1999; Gallup, 1993; Hill et al., 2000; Larson, Pattison, Blazer, Omran, & Kaplan, 1986; Shafranske, 1996). It has also been reported that a major source of personal strength for many people is drawn from their spiritual beliefs and practices (George, Larson, Koenig, & McCullough, 2000). Spirituality, however, still appears to lack the credibility needed for meaningful

research (Elkins, 1999). The field of psychology has produced very little research that supports the importance of spirituality from a clinical stance (Mack, 1994). It is believed by some that spirituality is an area of study that often fuels more controversy than valid research (Elkins, 1999). In fact, past decades of research have resulted in disagreement and confusion about what is actually meant by the term *spirituality* (Elkins, 1999; Hill et al., 2000; Zinnbauer et al., 1997). Regardless of the definition, however, the empirical research that has been conducted has repeatedly shown that a commitment to a spiritual relationship is usually associated with healthy physical, emotional, intellectual, occupational, and social functioning (Chandler et al., 1992; Ellison & Levin, 1998; Hawks, 1994; Maher & Hunt, 1993; Witmer & Sweeney, 1992).

A great majority of the literature in the area of spirituality focuses on the need for a comprehensive definition, instruments that are psychometrically sound, and researchers who are willing to embark on an area of controversial study.

The literature often refers to religion and spirituality as interchangeable constructs. Even the American Psychological Association's *Diagnostic and Statistical Manual of Mental Disorders* (1994) identifies

religion and spirituality problems under the same diagnosis code. The following criteria are classified under spiritual or religious problems: "distressing experiences that involve loss or guestioning of faith, problem association with conversion to a new faith, or questioning of spiritual values that may not necessarily be related to an organized church or religious institution" (p. 685). Research has continually defined both religion and spirituality largely in terms of religiosity (King & Dein, 1998; Mahoney & Graci, 1999 Westgate, 1996; Zinnbauer et al., 1997). Westgate (1996) postulates that this may exist because the construct religion is easier to operationalize. Researchers often look to frequency of religious attendance or acceptance of or satisfaction with certain religious beliefs for a more accurate measurement of the construct. In general, religion is often seen as a more narrow construct focusing on behavior, making it easier to define (Kroll, 1998; Westgate, 1996; Zinnbauer et al., 1997), whereas spirituality is often viewed as a broader, more abstract construct making it difficult to define. Chandler et al. (1992), however, suggest that spirituality is often independent of religion claiming that "spirituality can occur in or out of the context of religion, and not all aspects of religion are assumed to be spiritual" (p. 170).

Mahoney and Graci (1999) conducted a study that attempted to distinguish religiosity from spirituality. They posited that religiosity and spirituality were separate, contrasting constructs, rather than interchangeable. Further, they maintained that the meanings of these two terms change constantly. They drew upon leading authorities in the areas of death studies and spiritual studies to help examine the dimensions of spirituality and religiosity. The results of this study revealed several common themes among both groups of experts. Both considered themselves spiritual but not religious and were in general agreement that spiritual experiences were meaningful learning opportunities. Common themes most strongly associated with spirituality were charity, connectedness, compassion, forgiveness, hope, meaning, and morality. More than 80% in both groups believed that spirituality involved forgiveness. Nearly 90% of each group agreed that compassion was a dimension of spirituality. Further, spiritual individuals tended to be more hopeful and to experience more meaning or purpose in life than their nonspiritual or, in this case, religious peers. The authors suggested further research in the clarification of what people mean by spirituality and how it is expressed in daily life.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

In agreement with Mahoney and Graci, Pargament (1999) also sees the need for researchers to clearly identify what is meant by the terms spirituality and religiosity. In contrast, however, Pargament posits that although the literature appears to be attempting to distinguish religion from spirituality, he sees no need for the distinction between terms. In fact he claims that separating these concepts operationally would be impossible, largely because most individuals perceive religion and spirituality as encompassing similar components. Both focus on the sacred or divine, beliefs about the sacred, the effects of those beliefs on behavior, practices used to attain or enhance a sense of the sacred, and experiences of spiritual or religious states of consciousness. The major difference appears to be how these two terms are viewed and, thus, interpreted by researchers. Often religion is described as, and confined to, institutional settings, whereas, spirituality is often linked with more individualized religious expression. Pargament suggests that by categorizing the two constructs in this manner, the relationship between them may be threatened. He believes that, the difference between broadband religion and narrowband religion must be made clear; that religion is not a synonym for institution, dogma, or ritual; and that

religion is not a dry, static, dead-end construct. "We must make it clear that our field is vitally concerned with spirituality and all that matters sacred" (p. 15).

In a recent study by Zinnbauer et al. (1997), a group of participants was asked whether they considered themselves to be religious and not spiritual, spiritual and not religious, both religious and spiritual, or neither religious nor spiritual. The majority (74%) defined themselves as both religious and spiritual. Findings indicated that this group was more likely to view religiousness in a positive light, more likely to engage in traditional forms of worship such as church attendance and prayer, more likely to hold orthodox or traditional Christian beliefs, less likely to be independent from others, less likely to engage in group experiences related to spiritual growth, less likely to hold non-traditional "new age" beliefs, less likely to have had mystical experiences, and less likely to differentiate religiousness and spirituality as different and non-overlapping concepts, compared to those who identified themselves as "spiritual but not religious." This study supports Pargament's (1999) theory of no separation needed between terms. However, the findings also suggest that these particular participants are more characteristic of spiritual persons

than religious persons, as described by the current literature.

With due respect to Pargament's perspective, the need for a unified definition of spirituality appears to be overwhelming in the literature. The problem, however, is how to come to a unified definition with so many different concepts of spirituality. Many believe that there are several contributing factors which make this task insurmountable. Studies show that although Americans may desire spirituality, it may not be in the form of organized religion (McGovern, 1998; Zinnbauer et al., 1997). As stated earlier, spirituality is largely measured in terms of religious constructs. Therefore, theoretically, if people are not participating in organized religion, the measurement of spirituality becomes increasingly more difficult. Further, over the past several decades the interest in spirituality has appeared to increase, while reassurance in religion and religious leadership has decreased. Roof (1993) found in his study that baby boomers dropped out of organized religion in large numbers from the following religious persuasions: 84% were Jewish, 69% were mainline Protestants, 61% were conservative Protestants, and 67% were Catholic (Zinnbauer et al., 1997). Because their spiritual needs were not being met, many left the churches and

synagogues in search of a more individual or personal spiritual experience. Many of the non-traditional practices sought were: yoga groups, 12-step programs, meditation groups, Jungian psychology, Greek mythology, new age philosophies, and Eastern practices. Pargament (1999) posits that these are just new forms of spiritual institutions in the guise of healing groups. It appears that what they are truly in search of are places where they are able to share their views and receive support of other like-minded people. For some, this created a whole new dimension to spirituality and its meaning for many people. For others, it added validity to the concept that religion and spirituality are similar in nature and that it is the perception of the individual constructs that is relevant to meaningful religious or spiritual experiences.

Another prevalent area of concern is the lack of active contributors to research in the area of spirituality (Bergin & Jensen, 1990; Goodloe & Arreola, 1992; Hill et al., 2000; Shafranske, 1996). Goodloe and Arreola (1992) postulate several reasons as to why this may exist.

Spiritual health cannot be defined and researched and therefore, cannot be a legitimate area of study; spirituality is a part of the mental/emotional psychological dimension of health and does not need to be recognized as a separate entity within health education; and, recognition of spiritual health implies

the separation of church and state in education. (p. 222)

Shafranske (1996) posits, and is heavily supported by the literature (Bergin & Jensen, 1990; McGovern, 1998), that spirituality seems to play a minimal role in the lives of most psychologists in the United States. A study by Bergin and Jensen (1990) concluded that psychologists were the least religious of the major mental health provider groups. Only 33% of clinical and counseling psychologists described faith as the most important influence in their lives, as compared to 72% of the general population. Further, a 1989 Gallup poll found nearly 50% of psychiatrists and psychologists in the United States to be atheistic or agnostic, as compared to 6% of the general population. This same Gallup poll revealed that 66% of health professionals' raised in religious homes, abandoned their faith in adult years (McGovern, 1998). Thus, it could be surmised, that if religion plays an unimportant part in the lives of most researchers, it would most likely be an unimportant area of study for them.

Not all studies have painted such a negative picture of the relationship between religion and mental health. Religious beliefs and psychotherapeutic orientations were examined in a national sample of 237 clinical and counseling

a recognition of religiosity which evades the issue of

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

psychologists. Interestingly enough, findings in this particular study suggested American psychologists did take religion seriously and perceived it relevant both personally and professionally. Sixty-six percent believed in the transcendent, 72% believed their religious beliefs influenced their practice of psychotherapy, and 66% believed their practice of therapy influenced their religious beliefs (Bilgrave & Deluty, 1998).

There also appears to be a lack of interest in addressing spiritual issues and mental health in many training programs. A review of quantitative research studies published between 1991 and 1995 in three major mental health nursing journals revealed that only 5% of clinical psychologists and 30% of psychiatrists claimed to have had religious or spiritual issues addressed in their professional training programs (Weaver, Flannelly, Flannelly, Koeniq, & Larson, 1998; Larson, Larson, & Puchalski, 2000). Now more than 40 U.S. medical schools are incorporating faith in medicine in their school curriculums. As of January 1996, all psychiatric residency programs in the U.S. are required to address spiritual issues in their training programs. "Emphasis is on teaching medical students to inquire into the patient's belief system and see how it can enhance or slow the process of healing"

(McGovern, 1998, p. 23). This could be a giant leap for the field of psychology, as it appears that the medical model has been viewed by most researchers as the predominant model used and respected in the area of mental health research.

Witmer and Sweeney (1992) present a non-medical model that they believe represents the pathway to optimum health. They postulate 11 characteristics necessary for obtaining optimal health and wellness. These characteristics are grouped into five "life task" categories: (1) spirituality, (2) self-regulation, (3) work, (4) friendship, and (5) love.

The first life task, spirituality, is viewed by Witmer and Sweeney (1992) as the center of wholeness. It encompasses oneness and inner life, and purposiveness, optimism, and values. The second life task, self-regulation, includes such characteristics as sense of worth, sense of control, realistic beliefs, spontaneous and emotional responsiveness, intellectual stimulation, problem solving, and creativity; a sense of humor, and physical fitness and nutrition. The third life task, work, is a fundamental life task that provides psychological, social, and economic benefits to one's well-being. Friendship is the life task characterized through social interests and connectedness with others, outside of intimate relationships. The final life task is love.

It is important to note that Witmer and Sweeney (1992) view these forces as crucial components of optimum mental health, noting that spirituality is at the center of wholeness and the catalyst of well-being. They posit that these forces ultimately impinge on the health and well-being of every individual.

#### Attitude Toward Life

Growing evidence suggests that positive emotional states may influence mental health (Peterson, 2000; Salovey, Rothman, Detweiler, & Steward, 2000; Seligman, 1998), promoting healthy perceptions, beliefs, and physical wellbeing. Studies show that individuals who possess a positive outlook on life or a more optimistic attitude do better in their personal and professional lives than those who do not (Lee & Seligman, 1997; Peterson, 2000; Seligman, 1998). They are more apt to experience better health, less depression, higher achievement, better health habits, contentment, and a greater sense of coherence or resilience (Adams & Bezner, 2000; Dember & Brooks, 1989; Seligman, 1998). Optimism has been linked to positive mood and good morale; to perseverance and effective problem-solving; to academic, athletic, military, occupational, and political success; to

popularity; to good health; and to long life and freedom from trauma (Peterson, 2000).

The literature suggests that an optimistic attitude about the future may be health protective. Individuals with a positive sense of self-worth, belief in personal control, and an optimistic outlook appear to experience better health habits, find meaning in life, develop better coping skills, enhance social resources, and recognize the value of social relationships (Taylor, Kemeny, Bower, Gruenewald, & Reed, 2000). Seligman and Czikszentmihalyi(2000) posit that individuals today often lack the positive features that make life worth living: hope, wisdom, creativity, future mindedness, courage, spirituality, responsibility, and perseverence. They also suggest that society chooses to focus on the pathology of the individual rather than on the prevention. Their theory of Positive Psychology represents change in the perception of psychology, focusing on positive qualities of the individual as opposed to the preoccupation of repairing life's crises.

The psychological and emotional state of an individual appears to be important enough to sound mental health that the American Psychological Association's journal, American Psychologist (2000), devoted an entire special issue to this topic. Happiness, excellence, optimal human functioning,

subjective well-being, the future of optimism, selfdetermination, positive psychology, creativity, and positive youth development were only a few of the areas addressed. One particular author, Myers (2000), addressed the scientific pursuit of happiness. Researchers examined possible associations between happiness and (1) economic growth and personal income, (2) close relationships, and (3) religious faith. Myers revealed, through his review of the research, that age, gender, and income really produced very little insight into why people were happy, whereas a supportive network of close relationships and faith that encompassed social support, purpose, and hope did contribute to why people were happy. Myers suggested that a possible reason for the correlation between faith and well-being was the sense of meaning and purpose that was derived from one's faith. People of faith tended to retain or recover greater happiness after suffering divorce, unemployment, serious illness, or bereavement (Ellison & Smith, 1991; McIntosh, Silver, & Wortman, 1993).

Several studies have looked at optimism and pessimism as they relate particularly to mental health (Adams & Bezner, 2000; Dember & Brooks, 1989; Lee & Seligman, 1997; Reilley & Dember, 1998; Seligman, 1998).

Adams and Bezner (2000) conducted a study that explored the relationship between spiritual and psychological dimensions. Undergraduate college students were administered a series of survey instruments which resulted in several noteworthy implications. Optimism, a sense of coherence, and a meaningful purpose in life directly affected overall wellness. The findings suggested that an optimistic outlook and a sense of coherence must be present to enhance a sense of overall well-being. Further, the authors found that optimism and a sense of coherence might be internal resources that enable individuals to fulfill their own life purpose. They concluded that individuals with high levels of optimism and a strong sense of coherence possessed an enduring sense of personal control.

Dember and Brooks (1989) assessed the test-retest reliability of the Optimism/Pessimism Questionnaire and revealed several additional correlations in the study. One finding of particular interest was the correlation between optimism and pessimism scores and the two measures of happiness. It was revealed that people who reported themselves as content with their lives tended to be both high in optimism and low in pessimism.

Lee and Seligman (1997) studied optimism among White Americans, Chinese Americans, and Mainland Chinese.

Differences between the groups were studied with respect to success attribution. Among the findings, it was suggested that optimism was related to higher rates of academic and financial accomplishment, psychological confidence and persistence, and physical health among the Mainland Chinese.

It is worth noting that one's attitude toward life may very well impact one's mental health as well as one's spiritual beliefs, yet, scant data have been gathered that examine this relationship.

# Mental Health and Wellness

Recent empirical research (Chandler et al., 1992; Ellison & Levin, 1998; Hawks, 1994; Maher & Hunt, 1993; Marquand, 1995; Witmer & Sweeney, 1992) strongly suggests that a commitment to a spiritual relationship is usually associated with healthy physical, emotional, intellectual, occupational, and social functioning. According to Hettler (1979) these six major dimensions conceptualize wellness. Still, others claim that these dimensions must be in balance to achieve optimum wellness (Chandler et al., 1992; Hawks, 1994; Maher & Hunt, 1993; Witmer & Sweeney, 1992). Several wellness models even suggest that the spiritual dimension is the one component that is interactive with all the other dimensions (Chandler et al., 1992; Crose et al., 1992;

Goodloe & Arreola, 1992; Hettler, 1979; Witmer & Sweeney, 1992), stressing the importance of spirituality to mental health. Some go a step further to view spirituality as not only an important component to the balance, but at the very core of wellness (Goodloe & Arreola, 1992; Witmer & Sweeney, 1992). The literature suggests that the degree to which one is religiously committed is also important. One study, in particular, found that uncommitted believers suffered higher levels of anxiety than those who were either religiously committed or non-believers (McGovern, 1998). Religiously or spiritually involved persons have been shown to have fewer chronic conditions; lower functional disability; lower blood pressure; fewer strokes, less depression; anxiety, suicide, and alcoholism; higher life satisfaction; higher academic and work performance, and greater well-being (Chandler et al., 1992; Chirban, 1992; Ellison & Levin, 1998; Hawks, 1994; Koenig, 1995; Kroll, 1998; McGovern, 1998; Seligman, 1998; Thoresen, 1999). However, mixed opinions still exist on whether the relationship between religion and mental health is beneficial or detrimental to the individual. In a metaanalysis of results across studies, Bergin (1983) found that 23% of studies reported a negative relationship between religion and mental health, 47% reported a positive

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

relationship, and 30% reported no relationship. The majority of studies reflected a positive relationship between religion and mental health, again noting the importance of religion/spirituality in mental health.

Several studies also suggest a relationship between spiritual wellness and depression (Bilgrave & Deluty, 1998; Ellison & Levin, 1998; Koenig, George, & Peterson, 1998; Westgate, 1996). Depression appears to be one of the most common, more debilitating mental disorders of today, affecting anywhere from 5% to 30% of the U.S. population. Over the past decade adults have been living longer and more healthy lives, and it is expected that by the year 2030 older adults (85+) will make up 20% of the entire population (U.S. Bureau of the Census, 1993).

In a recent study, Koenig et al. (1998) interviewed 87 sick, elderly, depressed patients and found that the more religious patients recovered significantly faster from their depression. For every 10-point increase in religiosity, there was a 70% faster recovery (p. 540). Koenig (1994) suggests that religion and spirituality are particularly helpful in combating depression among the elderly. He posits that both religion and spirituality are particularly helpful to the elderly in that they provide meaning to death; hope at the end of the life cycle; and are effective

in coping with disability, illness, and negative life events. Studies are showing that physical health, specifically exercise, is only one dimension that researchers believe contributes to living longer, healthier lives. Healthy aging must also encompass the emotional, mental, social, spiritual, and environmental dimensions (Marinelli & Plummer, 1999).

As stated earlier, spirituality is often viewed as an abstract concept defined largely in terms of religious Thus, measurement of spirituality may be often construct. reliant on observable, measurable, religious behavior (i.e., church attendance, tithing). Gartner, Larson, and Allen's (1991) extensive review of approximately 200 studies on the relationship between religious commitment and psychopathology, indirectly addressed each of the abovementioned dimensions of wellness. The results of the review were divided into three sections: literature that suggested religion was associated with mental health, literature that suggested an ambiguous or complex relationship, and literature that suggested religion was associated with pathology (Shafranske, 1996). The first section discussed studies associated with physical health, mortality, suicide, drug use, alcohol abuse, delinguency and criminal behavior, divorce and marital satisfaction, well-being, outcome, and

depression. The second section included studies related to health, anxiety, psychosis, self-esteem, sexual disorders, intelligence/education, and prejudice. The final section presented studies that included authoritarianism, dogmatism, tolerance and rigidity, suggestibility and dependence, selfactualization, and temporal lobe epilepsy. Of the multitude of studies reviewed by Gartner et al. (1991), four trends evolved:

1. Most studies linking religious commitment to psychopathology employed mental health measures called soft variables (paper/pencil personality tests that measure theoretical constructs). In contrast, most research linking religion to positive mental health studied hard variables (real life behavioral events which can be observed and measured).

2. Low levels of religiosity are most often associated with disorders related to undercontrol of impulses, whereas high levels of religion are most often associated with disorders of overcontrol (p. 15)

3. Behavioral measures of religious participation are more powerfully associated with mental health than are attitudinal measures.

4. Distinctions such as that between intrinsic and extrinsic religion explain some inconsistent findings.

These finding may attribute to the figures revealed in Bergin's (1983) meta-analysis.

## CHAPTER III

### METHODOLOGY

The purpose of this study was to determine if a relationship could be found between spiritual belief, attitude toward life, and mental health among members of a physical fitness club in Northern Indiana. This chapter discusses the design of the study, the research sample, instrumentation, procedures, null hypotheses, and methods of analysis.

#### Design of the Study

This study employed the survey research method in which the data were gathered by using three surveys: the Royal Free Questionnaire on Beliefs and Experiences, the Optimism and Pessimism Questionnaire, and the Mental Health Inventory. These data were analyzed statistically using either multiple regression analysis, canonical correlation analysis, or t- test analysis to examine spiritual belief, attitude toward life, and mental health.

35

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

#### Sample

The sample for this study consisted of individuals who were members of a co-ed health club located in Northern Indiana with approximately 1,600 members. Both male and female participants were randomly selected from the database provided by the Northwest Athletic Club. This population was chosen for this study for several reasons: diversity in age (ranging from upper teens to 80+); good representation of both genders; members were not affiliated with any particular religious organization (thus, minimizing religious bias); membership number large enough to obtain a good sample; and the manager of the facility was willing to participate, making the population easily accessible. Also, the literature supports the concept of a natural relationship between the mind, body, and soul (Chandler et al., 1992; Ellison & Levin, 1998; Myers, 1992; Witmer & Sweeney, 1992), thus, making individuals willing to make the commitment toward good physical health emerge as a worthy selection for this study. Few studies, if any, have researched this particular population.

#### Instrumentation

A battery of three instruments was used in this study: the Royal Free Questionnaire on Beliefs and Experiences, the

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

Optimism and Pessimism Questionnaire, and the Mental Health Inventory (MHI).

# The Royal Free Questionnaire on Beliefs and Experiences

This instrument was developed by King, Speck, and Thomas in 1994 and has been refined twice since its development. Originally titled the Royal Free Interview for Religious and Spiritual Beliefs, it is now called the Royal Free Questionnaire on Beliefs and Experiences. It was first refined as a result of a study conducted at the Royal Free Hospital of London involving 300 patients. More recently, King states that they again "refined and extended the instrument a little, based on our experience with it" (M. King, personal communication, August 2000). The current form of the instrument is now in a questionnaire format. The questionnaire is being used in at least two other studies, neither of which have been published. One is a study on the outcome of bereavement and the other is a national study on ethnic minority mental health in Britain involving 7,000 to 10,000 people (M. King, personal communication, August 2000). The author hoped to submit both these studies for publication.

The Royal Free Questionnaire on Beliefs and Experiences is an 18-item instrument using a Semantic-Differential type

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

response format measuring spiritual belief. It begins by obtaining demographic information and a brief selfdescription from each individual. Within the instrument are instructions as well as definitions of religion and spiritual beliefs or experiences that will be used by the individual as he or she takes the survey. Some of the questions require individuals to respond quantitatively by indicating on an 11-point scale (0 - 10) how strongly he or she holds a particular belief (Hill & Hood, 1999). Cut-off scores were derived by dividing the 11-point scale roughly into thirds and placing the individual responses into three groupings. Low scores of 0, 1, 2, or 3 fell in the first grouping of weak spiritual beliefs. Moderate scores of 4, 5, or 6 fell in the second grouping of moderate spiritual beliefs. High scores of 7, 8, 9, or 10 fell in the third grouping of strong spiritual beliefs.

## Spiritual and Philosophical Scales

This questionnaire is composed of two scales: the spiritual scale, obtained from the sum answers to questions 3, 9, 10, 11, and 12, and the philosophical scale, obtained from the sum answers to questions 3, 15, 16, and 17. In the questionnaire format, the answers that sum the spiritual scale are 3, 8, 9, 10, and 11 instead of 3, 9, 10, 11, and

12. Only the sequencing of the questions changed, not the content (Hill & Hood, 1999).

### Reliability and Validity

Internal reliability was computed for the spiritual and philosophical scales. The alpha coefficient for the spiritual scale was 0.81, and 0.60 for the philosophical.

Test-retest reliability was measured by assessing 103 of the hospital staff subjects a second time 1 week later. Correlation coefficients for the questions scored using the 11-point scale were high and ranged from 0.76 to 0.93: five correlations were above .90, four were above .85, and four were below .85 (Hill & Hood, 1999).

Concurrent validity was unknown as comparisons with other measures of religious or spiritual belief were not made.

Criterion validity was assessed in two ways: (1) measuring the strength of the association between spiritual belief (spiritual scale score) and religious observance, and (2) comparing the spiritual scale score of a third reference group (the individuals strongly associated with a religious faith) with the spiritual scores of the other two reference groups (Hill & Hood, 1999).

The Pearson correlation for the association between the spiritual scores and religious observation was 0.41 (p < .0005). A linear relationship between frequency of religious observance and strength of belief was found (Hill & Hood, 1999).

The Optimism and Pessimism Questionnaire

The Optimism and Pessimism Questionnaire, also known as the Optimism and Pessimism Instrument, was developed by Dember, Martin, Hummer, Howe, and Melton in 1989. The instrument consists of 56 Likert-type items, each rated on a 4-point scale: (1) strongly agree, (2) disagree, (3) agree, and (4) strongly disagree. It is composed of two separate 18-item scales. One set measures optimism and implies an optimistic outlook, while the other set measures pessimism implying a pessimistic outlook. The items are reversescored with 20 filler items added to mask the intent of the instrument. Scores for each scale can range from 18 to 72 (Hummer & Dember, 1992; Lewis & Dember, 1995).

### Reliability and Validity

The two scales, optimism and pessimism, are only moderately correlated, with the value of r over a series of studies ranging from -.52 to -.65 (Dember & Brooks, 1989; Dember et al., 1989; McConnell & Bill, 1993).

Internal consistency is satisfactory, with Cronbach's alpha .86 for optimism and .88 for pessimism (Dember et al., 1989).

The test-retest reliability was .75 for optimism and .84 for pessimism after 2 weeks (Dember & Brooks, 1989).

Of the instruments available for measuring an optimistic attitude, the Optimism/Pessimism scale is the most recent instrument (Lewis & Dember, 1995), providing a global perspective of optimism and pessimism, both of which are considered an important attribute of human thought and expression (McConnell & Bill, 1993).

### Mental Health Inventory

The Mental Health Inventory is a 38-item survey instrument fielded by RAND's Health Insurance Experiment which focuses on psychological distress and psychological well-being. This instrument is self-administered by individuals 14 years of age or older. The items in the MHI yield six subscales: anxiety, depression, loss of behavioral/emotional control, general positive affect, emotional ties, and life satisfaction; two global scales: psychological well-being and psychological distress; and an overall mental health index. Eight additional items are interspersed among the MHI items because of similar content. Each of the 38 items asks about the frequency or intensity

of a psychological symptom during the past month. All but two items are accompanied by a six-choice response scale; two use 5-point response scales.

#### Reliability and Validity

The reliability of the mental health index scores, based on combined sites (*n* = 5089), was high at 0.96. Although the coefficients varied across the health insurance experiment sites, the lowest reliability estimate was 0.93. The stability of the mental health index has been estimated for intervals of 1, 2, and 3 years between administrations. One-year stability coefficients ranged from 0.60 to 0.76; 2year stability coefficients ranged from 0.54 to 0.69; and 3year stability coefficients from 0.54 to 0.58. validity.

## Mental Health Subscales

The mental health subscales are scored in two steps: (1) item scoring; and (2) the subscales themselves. Of the 38 mental health items, 35 are used to score the six mental health subscales (55, 80, and 99 are not included). Each item appears in only one subscale. Item scoring depends on two things: (1) whether the scores on the precoded-item responses indicate more frequent or intense occurrence or favorable or unfavorable mental health symptoms, and (2) whether the item belongs to a positively or negatively

scored mental health subscale. All subscales are scored so that higher scores indicate more of the construct named by the subscale's label. Thus, higher scores on three of the subscales indicate positive states of mental health: general positive effect, emotional ties, and life satisfaction. Higher scores on the other three subscales indicate negative states of mental health: anxiety, depression, and loss of behavioral/emotional control. The object of item scoring is to ensure that higher scores on each item reflect more of the construct named by the scale to which it belongs. Items are scored and then summed to calculate each subscale score.

#### Global Mental Health Scales

All 38 items are used to score the global mental health scales. The first global scale, psychological distress, contains 25 items. While 14 items are used to score the second global scale, psychological well-being. As stated earlier, item-scoring rules depend on, (1) whether the scores on the precoded-item responses indicate more frequent or intense occurrence or favorable or unfavorable mental health symptoms, and (2) whether the item belongs to a positively or negatively scored mental health subscale. Both global scales are scored so higher scores indicate more of the construct named by the scale's label. Higher scores

on psychological distress indicate negative states of mental health, whereas higher scores on the psychological wellbeing scale indicate positive states.

# Mental Health Index

All 38 mental health items are used to score the mental health index. Higher scores on each item reflect more frequent occurrence of favorable mental health symptoms, indicating greater psychological well-being. Thus, precoded responses for all favorably worded items on which higher precoded response scores indicate infrequent occurrence are recoded, as are all unfavorably worded items on which higher precoded response scores indicate frequent occurrence.

Item scores are summed to calculate the index score ranging from 38 to 226, with higher scores indicating an absence of psychological distress and frequent experiences of psychological well-being during the past month.

# Procedure

A list from the Northwest Athletic Club was obtained. Approximately 700 out of 1,600 individuals were randomly selected by the manager of this facility. The manager was asked to provide names and addresses, eliminating duplicate addresses and family members from the pool. Upon receiving

the list of names and addresses of the members, each individual was sent a packet containing the following: a cover letter from the researcher with instructions, the Royal Free Questionnaire on Beliefs and Experiences, the Optimistic/Pessimistic Questionnaire, and the Mental Health Inventory. The cover letter explained the purpose of the study, what respondents' involvement would be, and a note of appreciation for their participation. In an effort to elicit the maximum possible response rate, the first 200 individuals to return their packet were entered in a drawing for a 3-months-free membership. Each participant received a numbered ticket that corresponded with the number on their survey. They were asked to keep this ticket and return their completed surveys to the front desk at the Northwest Athletic Club. Two weeks later a follow-up letter was sent to those individuals who did not respond to the survey, along with a stamped, self-addressed envelope to my home address. Two individual ticket numbers were then chosen from the free membership offer on October 4, 2000, by the manager. The numbers were posted at the front desk, and the winners were to present their winning tickets within 5 days of the drawing. Only one member claimed his or her free membership.

The participants ranged in age from 18 to 84 years. Females represented 67% of the sample and males represented 33%. The majority of the participants were married (69%), while slightly more than 30% were either living with a partner, divorced, widowed, separated, or single. Nearly 95% were self-described as Caucasian, with the remaining 5% described themselves as either African American, Asian, Hispanic, or other. Slightly less than 80% were employed at the time of the survey, 10% were retired, and 10% were described as either a home manager, a student, or unemployed and seeking work.

### Null Hypotheses

The research questions for this study led to the following Null Hypotheses:

Research Question 1. Do people with higher levels of spirituality have a better attitude toward life?

Null Hypothesis 1: There is no relationship between spiritual belief and attitude toward life as measured by optimism and pessimism.

Research Question 2. Do people with higher levels of spirituality report better mental health?

Null Hypothesis 2: There is no relationship between spiritual belief and mental health.

Null Hypothesis 2a: There is no relationship between

spiritual belief and anxiety.

Null Hypothesis 2b: There is no relationship between spiritual belief and depression.

Null Hypothesis 2c: There is no relationship between spiritual belief and loss of behavioral/emotional control.

Null Hypothesis 2d: There is no relationship between spiritual belief and general positive affect.

Null Hypothesis 2e: There is no relationship between spiritual belief and emotional ties.

Null Hypothesis 2f: There is no relationship between spiritual belief and life satisfaction.

Null Hypothesis 2g: There is no relationship between spiritual belief and psychological distress.

Null Hypothesis 2h: There is no relationship between spiritual belief and psychological well-being.

Research Question 3. Is there a relationship between spiritual belief, attitude toward life, mental health, and age?

Null Hypothesis 3: There is no difference between spiritual belief, attitude toward life, mental health, and age.

Research Question 4. Is there a difference in spiritual belief, attitude toward life, and mental health between men and women?

Null Hypothesis 4: There is no difference in spiritual belief, attitude toward life, and mental health between men and women.

Research Question 5. Is there a linear relationship between spiritual belief, attitude toward life, and mental health?

Null Hypothesis 5: There is no significant canonical correlation between any linear combination of the six spiritual belief variables, the two attitude toward life variables, and the eight mental health variables.

### Data Analysis

Hypotheses 1 through 4 were tested by zero-order correlation analysis and regression analysis. In each instance, spiritual belief involved six variables, attitude toward life involved two variables, and mental health involved eight variables. Spiritual belief variables included: view of life, importance in practice of belief, influences of beliefs on world affairs, day-to-day life, natural disasters, and coping with life events. The following components of mental health were measured by the Mental Health Inventory: anxiety, depression, loss of emotional/behavioral control, general positive affect, emotional ties, and life satisfaction. Two additional components were measured also using the MHI: psychological distress and psychological well-being. Two variables, optimism and pessimism, were measured by the Optimism/Pessimism Scale. Hypothesis 5 was tested by canonical correlation analysis to determine the relationship between two sets of variables, spiritual belief and mental health. All five hypotheses offered several possible combinations of variables or sets of variables to be tested.

#### Summary

This chapter presented the research design, sample, instruments and procedures used, research questions, null hypotheses, and the statistical techniques used to test each hypothesis. It postulates the relationships between variables comprising spiritual beliefs, attitude toward life, and mental health as measured by the Spiritual Belief Scale, Optimism/Pessimism Scale, and the Mental Health Inventory. This chapter presents the statistical analyses performed for each hypothesis.

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

### CHAPTER IV

# PRESENTATION AND ANALYSIS OF DATA

The primary purpose of this study was to examine the relationships between spirituality, attitude toward life, and mental health. This chapter presents the findings from the analyses of the sample data. The information is organized into three sections: (1) description of the participants in the study, (2) descriptive results of the survey, and (3) statistical analyses of the tested hypotheses.

## Description of the Sample

A total of 690 surveys were randomly distributed to both male and female members of a Northwest Indiana health club located approximately 60 miles east of Chicago situated in a town of 35,000 people. Of the 690 surveys distributed, 187 (27%) were returned and all were used in the analysis of the data. The participants' range of age was 18 to 84 years (Table 1). Females represented 66% of the sample and males represented 33%. The majority of the participants were married (69%), while slightly more than 30% were either

living with a partner, divorced, widowed, separated, or single. Nearly 95% were self-described as Caucasian, with the remaining 5% described as either African American, Asian, Hispanic, or other. Slightly less than 80% were employed at the time of the survey, 10% were retired, and 10% were described as either a home manager, a student, or unemployed and seeking work (Table 2). Eighty individuals (43%) reported having a religious understanding of their life, 47 (25%) reported a spiritual understanding of life, 48 (26%) reported both a religious and a spiritual understanding, and 11 (6%) reported neither a religious understanding nor a spiritual understanding of life (Table 3). Forty-eight percent of the respondents reported their faith as Catholic; 16% were either Methodist, Lutheran, Jewish, Baptist, or Buddhist; 6% were Presbyterian, and 30% represented the "other" category (Table 4).

Table 1

Gender	18	18 - 39		40-59		60 +		Total	
	Ν	00	N	υYo	Ν	o¥.	N	ofto	
Men	18	27.69	29	32.22	14	48.28	61	32.62	
Women	47	72.30	61	67.78	15	51.72	123	65.78	
Total	65	34.76	90	48.65	29	15.51	184	100.00	
Note. T	hree	female re	spond	lents did	l not	give age	e		

Percentage of Participants by Age and Gender

Table 2

Demographics of the Participants

Variable	n	- %	
Marital Status			
Married	129	68.98	
Living with partner	7	3.74	
Divorced	10	5.35	
Widowed	4	2.16	
Separated	3	1.62	
Single	32	17.30	
Missing	2	1.08	
Race			
Caucasian	177	94.65	
African American	2	1.08	
Hispanic	1	0.05	
Other	4	2.16	
Missing	3	1.62	
Employment			
Employed	149	79.68	
Unemployed	1	0.05	
Student	4	2.16	
Retired	19	10.16	
Home Manager	14	7.48	
Total	187	100.00	

Table 3

Distribution of Religious and Spiritual Understandings of Life

Variable	п	olo
Religious understanding	80	43.01
Spiritual understanding	47	25.26
Neither religious nor spiritual understanding	11	5.91
Both religious and spiritual understanding	48	25.81
Total sample	186	100.00

Table 4

Frequency Distribution of Religious Affiliations of the Respondents

Variable	п	00
Catholic	82	43.85
Other	51	27.27
Methodist	11	5.88
Baptist	10	5.35
Presbyterian	10	5.35
Buddhist	3	1.60
Judaism	3	1.60
No response	16	8.56
Total	187	100.00

# Descriptive Results of Survey

Both males and females reported high levels of spirituality. Table 5 presents the percentages of levels of spiritual beliefs among participants. Of the 187 respondents, 172 (92%) reported moderate or high levels of spiritual belief, while 15 (8%) reported low levels of spiritual beliefs (see p. 36). As reflected in Table 6, 157 (83.95%) respondents stated they prayed either alone, with others, or both. Slightly less than 50% participated in some type of ceremony, while slightly more than 50% reported that meditation was a part of their lives, either alone, with others, or both. One hundred and eight individuals (60%) stated they read and studied as part of practicing their faith, while 113 of those who responded (63%) claimed that contact with a religious leader did not necessarily play a part in their beliefs.

Additionally, 127 (70%) of those who responded reported that they communicated with a spiritual power (prayer or via a medium), 130 (70%) believed in life after death, and 89 (49%) had an intense experience in which they felt some deep meaning in life. An almost equal number of individuals, 69 (38%), reported never having had an intense experience in which they felt some deep new meaning in life. For those

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

individuals who did report this intense experience, it occurred more than once for 55 (60%) of them, only once for 36 of them (40%), and 96 (51%) did not report the frequency of their experience. This intense experience in which they felt some deep new meaning in life lasted only minutes (54%) for the majority of the participants. It lasted only seconds for 14 respondents (16%), hours for 11 respondents (13%), and days for only 15 (17%). One hundred and sixtyfour respondents (93%) reported never having had an intense experience described as a near-death experience (Table 7).

Table 8 presents the means, standard deviations, and scale ranges for the different measures of spiritual belief and mental health scales. It can be observed that the mean scores for view of life, influence in day-to-day-life, ability to cope with life events, and importance in practicing beliefs were 7.94, 7.82, 8.00, and 7.28, respectively. These participants embodied a strong belief in religious or spiritual views of life; in spiritual influence on daily life; in spiritual power that helps one cope with personal events in life; and in the importance of practicing beliefs. Belief that a spiritual force or power influenced world affairs or natural disasters was moderate

among participants, with mean scores of 6.07 and 5.68, respectively.

Table 5

Level of Spiritual Belief of Participants

Belief	n	ę		
High	83	44.39		
Moderate	89	47.59		
Low	15	8.02		
Total	187	100.00		

## Table 6

# Variables That Play a Role in Belief

Variable	Alone		1	With		Both	Ne	ither
N (187)			0	Others				
	N	oto	N	cyo	N	90	N	
Prayer	50	27.78	8	4.44	99	55.00	23	12.78
Ceremony	4	2.22	60	33.33	18	10.00	98	54.44
Meditation	70	38.89	1	.56	26	14.44	83	46.11
Reading	43	23.89	19	10.56	46	25.56	72	40.00
Contact religious leader	10	5.56	34	18.89	22	12.22	113	62.78

Note. Missing cases = 4%

Frequency Distribution of Questions Not Included in Total Spiritual Belief Scale

Variable	Near Death Intense Exp.		-	Frequency of Intense Exp.		ngth of ense Exp.
	N		N		N	<u>-</u> &
None	164	93				
Only Once			36	39.56		
> Than Once			55	60.44		
Seconds					14	16.09
Minutes					47	54.02
Hours					11	12.64
Days					15	17.24

Variable	Mean	SD	Scale Ranges
	Mental	Health	
Satisfaction	4.34	1.08	1 - 6
Depression	8.05	3.00	4 - 24
Emotional Ties	8.80	2.65	3 - 12
Loss	16.07	6.54	9 - 63
Anxiety	19.33	6.53	9 - 63
Affect	41.20	8.52	10 - 10
Psych Distress	48.60	15.75	24 -104
Well-being	58.49	11.91	14 - 84
	Spiritual	Beliefs	
View of Life	7.94	2.27	0 - 10
Day-to-Day	7.82	2.69	0 - 10
Cope w/ Events	8.00	2.64	0 - 10
World Affairs	6.07	3.26	0 - 10
Natural Dis	5.68	3.54	0 - 10
Practice Faith	7.28	2.95	0 - 10

Mean Scores, Standard Deviations, and Scale Ranges for Spiritual Belief Scale and Mental Health Scales

*Note.* n = 187

The participants, in the month preceding the study, reported being generally happy, satisfied, or pleased with their personal life. They reported being involved in full and complete relationships a good bit of the time. Low levels of depression, and infrequent mood swings, low spirits, and downheartedness were also characteristic of the participants during the past month. Respondents reported being emotionally stable most of the time, in firm control of their behavior, thoughts, and feelings. Mean scores also revealed low levels of anxiety among participants. They almost never experienced nervousness when faced with excitement of unexpected situations; were almost never rattled, flustered, or upset; and were only a little restless, fidgety, impatient, anxious, or worried during the past month. Respondents reported that the future looked hopeful and promising and expected to have interesting days more often than not. A good bit of the time participants were relaxed and tension free; generally enjoyed the things they did; and viewed life as a wonderful adventure. Most days, during the past month, participants woke up feeling fresh and rested. Little psychological distress was reported among participants during the past month. They reported feeling strained, stressed, pressured, or lonely,

only a little of the time. The psychological well-being of the participants was strong, reporting that most of the time they were able to relax without difficulty. The exercise at the club could have aided in the reduction of anxiety, tension, and stress.

The zero-order intercorrelation matrix (Table 9) shows several significant correlations. View of life or the strength of one's commitment to a particular belief system correlated positively with general positive affect, emotional ties, and psychological well-being. View of life correlated negatively with depression, loss of emotional/behavioral control, and psychological distress. The importance of practicing one's spiritual beliefs correlated positively with general positive affect, emotional ties, and psychological well-being. The level of belief in a spiritual force that influences daily decisions and world affairs resulted in a positive correlation with emotional ties. The level of belief in a spiritual force that enables one to cope with what life bestows correlated positively with general positive affect, and psychological well-being. Overall, individuals with a strong spiritual view of life were more likely to experience better mental health.

Inter-correlation Matrix Between the Six Spiritual Belief Variables, Two Life Attitude Variables, and Eight Mental Health Variables

Var	iable	3	7	8	9	10	11	A	D	L	AF	Е	S	DP	0	Р	WB	
3.	View	1.00				·	<u>-</u>											
7.	Belief	.71	1.00															
8.	Day-to-day	.67*	.66*	1.00														
9.	Cope	.73*	.77•	.80*	1.00													
10.	World Aff	.42*	.45*	.61*	.60*	1.00												
11.	Disaster	.35*	.39*	.53*	.49*	.75*	1.00											
A.	Anxiety	11	10	.04	. 01	.09	.08	1.00										
D.	Depression	19*	11	. 02	.02	.04	. 07	.70	1.00									0  -
L.	Loss	22*	18*	. 02	.01	.08	.10	.63	.73	1.00								
AF.	Affect	.32*	.28*	.09	.19*	.08	.07	53	69	60	1.00							
Ε.	Emotion	.25*	.25*	.20*	.19*	.15*	.10	30	45	40	. 59	1.00						
s.	Satisfaction	n .11	.11	04	.03	01	02	-,38	60	49	.70	.63	1.00					
DP.	Distress	19*	15	.04	.01	.08	.10	.89	. 86	. 90	67	43	55	1.00				
о.	Optimism	.21*	.16	.09	.15*	.05	.12	23	33	31	.52	. 21	. 39	33	1.00			
Ρ.	Pessimism	22*	19*	06	12	01	01	.43	.50	.47	59	36	42	.53	61	1.00		
WB.	Well-being	. 32*	.29	. 13	.21•	.09	.07	-,54	71	61	. 97	.75	.77	69	.48	57	1.00	

\*Significant at the .05 level.

There were significant correlations between optimism and pessimism and several spiritual belief variables. View of life, practice of faith, and belief in a spiritual force that influences one's ability to cope with life events positively correlated with optimism while negatively correlating with pessimism. The stronger one's view of life, practice of faith, and spiritual influences, the more likely respondents were to have an optimistic attitude.

## Statistical Analysis for Hypothesis Testing

Five hypotheses were formed for this study in an effort to explore the relationship between spiritual beliefs, attitude toward life and mental health. The statistical results appearing in this section of the research relate to the five hypotheses forming the basis of this study. Direct-method multiple regression analyses were used to examine the relationships between the variables in Hypotheses 1 through 4. Hypothesis 5 sought to establish the relationship between spiritual beliefs, attitude toward life, and mental health variables. Canonical correlation analysis was used for testing this hypothesis. Significance was set at the .05 level.

The following null hypotheses looked at the individual scores of the Optimism/Pessimism Instrument and their

relationships with the six independent variables used to measure spiritual belief.

Null Hypothesis 1 states: There is no relationship between spiritual belief and attitude toward life, as measured by optimism and pessimism.

A direct-method regression analysis was used to determine the relationship between optimism and six measures of spiritual beliefs: view of life, importance in practice of belief, influences of beliefs on world affairs, day-today life, natural disasters, and coping with life events. Table 10 shows the results of regression analysis. The six measures of spiritual beliefs accounted for 7% of the variance in optimism. At an alpha level of 0.05, this was not significant. Thus, no statistically significant linear relationship was found between optimism and the six measures of spiritual beliefs (F(6,145) = 1.828, p=0.098). Based on the analysis, the hypotheses was retained.

Variable	B	SE	Beta	t	Sig.
(constant)	(51.434)	(1.855)	,		
View of life	.582	.345	.220	1.688	.094
Practice of belief	.189	.262	.096	.720	.473
Day-to-day	431	.323	192	-1.335	.184
Cope events in life	.140	.384	.061	.366	.715
World affairs	321	.237	178	-1.358	.176
Natural disaster	.330	.201	.202	1.644	.102

Multiple Regression Analysis of Spiritual Beliefs on Optimism

Note.  $R^2 = 0.070$ ; F(6, 145) = 1.828; p = 0.098.

A direct method regression analysis was used to determine the relationship between pessimism and six measures of spiritual beliefs: view of life, importance in practice of belief, influences of beliefs on world affairs, day-to-day life, natural disasters, and coping with life events. The results presented in Table 11 show this regression analysis. The six measures of spiritual beliefs accounted for less than 8% of the variance in pessimism. At an alpha level of 0.05, this was not significant. Thus, no statistically significant relationship was found between pessimism and the six measures of spiritual beliefs (F(6,138) = 1.863, p=0.091). Based on the analysis, the hypotheses was retained. It is interesting to note, however, that view of life, defined as the strength of one's commitment to a particular spiritual belief system, was a significant predictor in the model (B=-.286). The weaker the view of life, the more negative the state of mental health appeared to be for the participant.

Null Hypothesis 2 states: There is no relationship between spiritual belief and mental health.

There was some indication that spiritual beliefs affected mental health. Significant correlations were observed between spiritual belief and depression, loss of behavioral/emotional control, general positive affect, emotional ties, psychological distress, and psychological well-being.

Variable	В	SE	Beta	t	Sig.
(constant)	(21.125)	(2.041)			
View of life	865	.370	286	-2.337	.021
Practice of					
belief	257	.284	115	904	.367
Day-to-day	.507	.357	.202	1.421	.157
Cope events in life	.247	.417	.095	.594	.554
World affairs	.128	.257	.063	.497	.620
Natural disaster	.054	.218	.029	.246	.806
11	B/C 130) 1	0.60			

Multiple Regression Analysis of Spiritual Beliefs on Pessimism

Note.  $R^2=0.075$ , F(6, 138)=1.863, p=0.091.

Sub null hypothesis 2a states: There is no relationship between spiritual belief and anxiety.

To determine the relationship between one's spiritual belief and anxiety, multiple regression was used. The results in Table 12 show that the predictor variables were not significant predictors of anxiety. The six measures of spiritual beliefs, as a set, explained only 7% of the variance in the criterion variable, anxiety. Based on this analysis, the null hypothesis was retained.

Variable	B	SE	Beta	t	Sig.
(constant)	(36.825)	(2.311)			
View of life	751	.416	230	-1.807	.073
Practice of					
belief	579	.316	241	-1.831	.069
Day-to-day	.273	.388	.101	.704	.483
Cope events					
in life	.467	.464	.166	1.006	.316
World affairs	.141	.294	.065	.481	.631
Natural disaster	.036	.260	.002	.014	.989

Multiple Regression Analysis of Spiritual Beliefs on Anxiety

Note.  $R^2=0.071$ ; F(6,153)=1.946; p=0.077.

Sub null hypothesis 2b states: There is no relationship between spiritual belief and depression.

Multiple regression analysis was used to determine the relationship between spiritual beliefs and depression. As a set, the six measures of spiritual beliefs were a significant predictor of depression. The measures accounted for slightly more than 10% of the variance in depression. This was significant at the .05 level (F(6, 154 = 3.04, p=0.008)). View of life was the most important predictor in the regression model (B=-.378), followed by coping with live events, approaching significance at .054. A stronger view

of life resulted in a less depressive state of mental health. There was a statistically significant relationship found between depression and the six measures of spiritual beliefs (Table 13). Thus, the null hypothesis was rejected.

Table 13

Variable	В	SE	Beta	t	Sig.
(constant)	(9.092)	(.887)			
View of life	506	.161	378	-3.143	.002
Practice of belief	166	.124	168	-1.345	.180
Day-to-day	.138	.155	.123	.888	.376
Cope events in life	.352	.181	.305	1.945	.054
World affairs	034	.112	038	308	.759
Natural disaster	.077	.095	.095	.810	.419

Multiple Regression Analysis of Spiritual Beliefs on Depression

Note.  $R^2 = 0.106$ ; F(6, 154) = 3.040; p = 0.008.

Sub null hypothesis 2c states: There is no relationship between spiritual belief and loss of behavioral/emotional control.

Table 14 shows the results of the regression analysis used to determine the relationships between loss of behavioral/emotional control and spiritual belief. The six measures of spiritual belief accounted for nearly 18% of the variance in loss of behavioral/emotional control. At an alpha level of 0.05, this was significant. There were three significant predictors within the regression model. View of life was the most important predictor in this model (B=-.445), followed by coping with life events (B=.351) and practice of belief (B=-.259). These results suggested that weak spiritual view of life, disbelief that a spiritual force enables one to cope with life events, and the lack of importance in practicing beliefs were relevant predictors of loss of behavioral or emotional control. Based on the analysis, the null hypothesis was rejected.

Table 14

(constant)       (19.801)       (1.981)         View of life       -1.378       .355      445       -3.877         Practice of      594       .274      259       -2.172       .         belief       Day-to-day       .373       .342       .145       1.091       .         Cope events       .936       .400       .351       2.339       .         World affairs       .010       .246      005       .043       .         Natural       .202       .209       .108       .964       .		<u>-</u>				
<pre>View of life -1.378 .355445 -3.877 . Practice of594 .274259 -2.172 . belief Day-to-day .373 .342 .145 1.091 . Cope events in life .936 .400 .351 2.339 . World affairs .010 .246005 .043 . Natural .202 .209 .108 .964 .</pre>	Variable	В	SE	Beta	t	Sig.
Practice of belief      594       .274      259       -2.172       .         Day-to-day       .373       .342       .145       1.091       .         Cope events in life       .936       .400       .351       2.339       .         World affairs       .010       .246      005       .043       .         Natural       .202       .209       .108       .964       .	(constant)	(19.801)	(1.981)			
belief Day-to-day .373 .342 .145 1.091 . Cope events in life .936 .400 .351 2.339 . World affairs .010 .246005 .043 . Natural .202 .209 .108 .964 .	View of life	-1.378	.355	445	-3.877	.000
Cope events         in life       .936       .400       .351       2.339       .         World affairs       .010       .246      005       .043       .         Natural       .202       .209       .108       .964       .		594	.274	259	-2.172	.031
in life.936.400.3512.339.World affairs.010.246005.043.Natural.202.209.108.964.	Day-to-day	.373	.342	.145	1.091	.277
Natural .202 .209 .108 .964 .	-	.936	.400	.351	2.339	.021
	World affairs	.010	.246	005	.043	.966
disaster	Natural disaster	.202	.209	.108	.964	.336

Multiple Regression Analysis of Spiritual Beliefs on Loss of Behavioral/Emotional Control

Note.  $R^2=0.175$ ; F(6,153)=5.391; p=0.000.

Sub null hypothesis 2d states: There is no relationship between spiritual belief and general positive affect.

Table 15 presents the relationship between general positive affect and spiritual beliefs. The six measures of spiritual beliefs accounted for slightly more than 15% of the variance in general positive affect. This was significant at the 0.05 alpha level. Thus a statistically significant relationship was found between general positive affect and the six measures of spiritual beliefs (F(6, 152) = 4.623, p=0.000). Two independent variables were significant predictors within this model; view of life, the more important of the predictors (B=.421), followed by dayto-day spiritual influence on life (B=-.264). View of life was of moderate importance in predicting general positive affect, defined as an emotion or feeling distinguished from cognition, thought, or action, while belief that a spiritual force or power influences what happens in day-to-day life was of low importance in predicting general positive affect. The null hypothesis was rejected.

Variable	В	SE	Beta	t	Sig.
(constant)	(32.961)	(2.551)			
View of life	1.630	.456	.421	3.576	.000
Practice of belief	.439	.375	.152	1.170	.244
Day-to-day	849	.429	264	-1.981	.049
Cope events in life	099	.509	030	195	.845
World affairs	083	.311	032	266	.790
Natural disaster	.059	.265	.025	.222	.825
	<b>F</b> (6, 150)				

Multiple Regression Analysis of Spiritual Beliefs on General Positive Affect

Note.  $R^2=0.154$ ; F(6, 152)=4.623; p=0.000.

Sub null hypothesis 2e states: There is no relationship between spiritual belief and emotional ties.

Table 16 shows the results of regression analysis used to determine the relationships between spiritual belief and emotional ties. The six measures of spiritual belief accounted for only 8% of the variance in emotional ties. At an alpha level of .05, this was significant. Thus, there was a significant relationship found between emotional ties and the six measures of spiritual beliefs (F(6,155)=2.243,p=0.042). However, though the set of predictors was significant, no significant individual predictors were revealed. This may be in part due to the strong intercorrelations among the predictor variables, shown in Table 9. Based on the analysis, the null hypothesis was rejected.

Table 16

		· · · · · · · · · · · · · · · · · · ·			
Variable	В	SE	Beta	t	Sig.
(constant)	(6.679)	(.801)			
View of life	.266	.145	.222	1.827	.070
Practice of belief	.147	.112	.167	1.321	.189
Day-to-day	.073	.140	.074	.522	.602
Cope events					
in life	209	.164	202	-1.274	.205
World affairs	.058	.101	.073	.578	.564
Natural disaster	042	.086	059	498	.619

Multiple Regression Analysis of Spiritual Beliefs on Emotional Ties

Note.  $R^2=0.080$ ; F(6,155)=2.243; p=0.042.

Sub null hypothesis 2f states: There is no relationship between spiritual belief and life satisfaction.

Multiple regression analysis was used to determine the relationship between spiritual belief, defined as a culmination of one's convictions and perceptions gained through one's spiritual and life experiences, and life satisfaction. The analysis, presented in Table 17, showed no significant relationship between life satisfaction and spiritual belief. The set of variables explained only 4% of the variance in the criterion variable life satisfaction (F(6,153)=1.249, p=.285. Therefore, the null hypothesis was retained.

Table 17

Multiple Regression Analysis of Spiritual Beliefs on Life Satisfaction

Variable	В	SE	Beta	t	Sig.
(constant)	(4.035)	(.332)		<u> </u>	
View of life	.116	.063	.237	1.840	.068
Practice of belief	.043	.063	.121	.933	.352
Day-to-day	080	.058	200	-1.377	.170
Cope events in life	028	.069	068	412	.681
World affairs	032	.042	010	077	.939
Natural disaster	064	.035	022	182	.856

Note.  $R^2 = 0.047$ , F(6, 153) = 1.249, p = 0.285.

Sub null hypothesis 2g states: There is no relationship between spiritual belief and psychological distress. A direct-method regression analysis was used to determine the relationships between psychological distress and the six measures of spiritual beliefs. The results of the analysis are presented in Table 18. The six measures of spiritual beliefs accounted for 14% of the variance in psychological distress. This was significant at the 0.05 alpha level. Therefore, a statistically significant relationship was found between psychological distress and the six measures of spiritual beliefs (F(6,151)=4.170,p=001). View of life resulted in the strongest predictor of psychological distress. Participants who held strong views of life revealed less psychological distress or more positive states of mental health. Based on the analysis, the null hypothesis was rejected.

Sub null hypothesis 2h states: There is no relationship between spiritual belief and psychological well-being.

Table 19 presents the regression analysis for psychological well-being and spiritual beliefs As a set, the six predictors of spiritual beliefs were a significant predictor of psychological well-being. The measures accounted for slightly more than 13% of the variance in psychological well-being. This was significant at the 0.05 level (F(6,150)=3.872, p=0.001). View of life was the only

significant predictor in the model. The stronger the view of life, the more positive the state of mental health for the participant. Based on the analysis, the null hypothesis was rejected.

Table 18

Variable	В	SE	Beta	t	Sig.
(constant)	(55.771)	(4.787)		<u> </u>	
View of life	-3.009	.858	411	-3.509	.001
Practice of belief	-1.117	.660	206	-1.691	.093
Day-to-day	1.098	.826	.180	1.329	.186
Cope events in life	1.688	.966	.268	1.748	.082
World affairs	.130	.594	.027	.218	.828
Natural disaster	.353	.505	.080	.699	.486

Multiple Regression Analysis of Spiritual Beliefs on Psychological Distress

<u>Note</u>.  $\underline{R}^2 = 0.142$ ,  $\underline{F}(6, 151) = 4.170$ ,  $\underline{p} = 0.001$ .

Variable	В	SE	Beta	t	Sig.
(constant)	(141.638)	(3.415)			
View of life	1.888	3.415	.366	2.974	.003
Practice of belief	.699	.502	.185	1.395	.165
Day-to-day	797	.576	189	-1.383	.169
Cope events in life	263	.694	060	380	.705
World affairs	.036	.418	.011	.087	.931
Natural disaster	106	.352	035	301	.763

Multiple Regression Analysis of Spiritual Beliefs on Psychological Well-Being

Note.  $R^2=0.134$ , F(6, 150 = 3.872, p=0.001.

Null Hypothesis 3 states: There is no significant relationship between spiritual belief, attitude toward life, mental health, and age.

A direct-method regression analysis was used to determine the relationship between spiritual belief, attitude toward life, and mental health, among age. Because the analysis of independent variables for spiritual belief, attitude toward life, and mental health, was explored in hypotheses 1 and 2, hypothesis 3 looks at the global score of spiritual belief and its relationship with attitude, mental health, and age. The results presented in Table 20 show the analysis. Spiritual belief, attitude toward life, and mental health, as a set, were significantly related to age. The linear combination among these variables accounted for slightly over 21% of the variance in age (F(11,128)=3.155, p=0.000). Specific predictors include anxiety (B=2.82), loss of emotional/behavioral control (B=3.01), and psychological distress (B=-2.80). The older the subject, the more likely they are to experience higher levels of anxiety, higher levels of loss of emotional and behavioral control, and lower levels of psychological distress. Therefore, the null hypothesis was rejected.

Null Hypothesis 4 states: There is no difference in spiritual belief, attitude toward life, and mental health between men and women.

An independent t-test was used to determine the difference between six measures of spiritual belief: view of life, importance in practice of belief, influences of beliefs on world affairs, day-to-day life, natural disasters, and coping with life events, and attitude toward life: optimism and pessimism, among men and women. Table 21 shows the results of the analysis. Females were slightly higher than males in all measures excluding one, pessimism. At the significance level of .05, females seem to have a slightly greater belief in a spiritual power or force that

influences what happens in day-to-day life. They also appear to embrace the idea that a spiritual power or force enables one to better cope with personal life events.

## Table 20

Multiple Regression Analysis of Spiritual Beliefs, Attitude Toward Life, and Mental Health among Age

Variable	В	SE	Beta	t	Sig.
(constant)	(70.176)	(19.904)	.125	1.451	.150
Spirit	.153	.105	058	546	.586
Optimism	140	.257	.014	.131	.896
Pessimism	.027	.209	1.401	2.768	.007
Anxiety	2.818	1.018	.397	1.623	.107
Depression	1.749	1.078	1.520	2.970	.004
Loss	3.009	1.013	144	179	.859
Affect	228-	1.278	084	308	.759
Emotion	.437	1.418	299	-1.915	.058
Satisfaction	-3.973	2.075	-3.316	-3.073	.003
Psych Distres	-2.797	.910	.362	.344	.732
Well-being	.415	1.207			

Note.  $R^2 = 0.213$ , F(11, 128) = 3.155, p = 0.000.

Variable	M	SD	M	SD	t	df	р
View of life	8.21	2.10	7.39	2.52	-2.29	176	.067
Practice of belief	7.58	2.86	6.62	3.06	-1.97	166	.517
Day-to-day	8.19	2.43	7.02	3.07	-2.74	176	.033*
Cope events in life	8.87	6.01	7.21	2.87	-2.84	178	.045*
World affairs	6.21	3.15	5.76	3.49	88	177	.224
Natural disaster	5.90	3.51	5.22	3.60	-1.20	177	.826
Optimism	55.14	5.81	54.42	6.18	75	173	.723
Pessimism	33.55	7.03	33.76	7.20	.18	162	.972

Mean Levels and Standard Deviations of Spiritual Beliefs and Attitude by Gender

\*Significant at the .05 level.

Table 22 shows the differences between men and women on the mental health subscales. These results indicated that only two variables, psychological distress and psychological well-being, differentiated the two groups significantly. Females had significantly higher psychological distress than males. Perhaps, for this reason, females had significantly lower levels of psychological well-being than males.

From these scores it would appear that the group participants were a fairly homogenous group. The majority

of the sample were happy most of the time, psychologically fit, possessed high general positive affect, and had a spiritual or religious understanding of their life.

Table 22

Mean Levels and Standard Deviations of Mental Health Variables by Gender

М	SD	М	SD	t	df p
20.32	6.72	17.22	5.59	-3.08	182 .085
8.56	3.07	6.98	2.55	-3.45	183 .344
17.24	7.02	13.63	4.59	-3.62	182 .070
40.54	8.87	42.53	7.65	1.49	182 .059
8.85	2.67	8.68	2.61	41	185 .888
4.28	1.11	4.46	1.01	1.05	182 .296
51.38	16.29	42.68	12.76	-3.58	179 .024
152.02	11.78	154.47	9.75	1.51	179 .031
	20.32 8.56 17.24 40.54 8.85 4.28 51.38	20.32       6.72         8.56       3.07         17.24       7.02         40.54       8.87         8.85       2.67         4.28       1.11         51.38       16.29	20.326.7217.228.563.076.9817.247.0213.6340.548.8742.538.852.678.684.281.114.4651.3816.2942.68	20.326.7217.225.598.563.076.982.5517.247.0213.634.5940.548.8742.537.658.852.678.682.614.281.114.461.0151.3816.2942.6812.76	20.32       6.72       17.22       5.59       -3.08         8.56       3.07       6.98       2.55       -3.45         17.24       7.02       13.63       4.59       -3.62         40.54       8.87       42.53       7.65       1.49         8.85       2.67       8.68       2.61      41         4.28       1.11       4.46       1.01       1.05         51.38       16.29       42.68       12.76       -3.58

Note. Female n=123; male n=61.

\*Significant at the .05 level.

Null Hypothesis 5 states: There is no significant canonical correlation between any linear combination of the six spiritual belief variables, the two attitude toward life variables, and the eight mental health variables. The first function was found to be statistically significant at the .05 level. This function yielded a canonical correlation of r = .487 ( $R^2 = .238$ ), with  $X^2 =$ 612.82 and p<.0073. The null hypothesis was rejected. Table 23 presents the canonical variable loadings in the two sets of variables for this function. Determining the cutoff for interpretation of the loadings is based on personal preference. However, it is common procedure to initially note only variables whose weights are as least half the value of the greatest weight. These variables are ranked and considered significant in the canonical analysis. Variables with loadings below .30 are usually not interpreted (Tabachnick & Fidell, 1996).

As indicated in Table 23, the significant canonical function suggested that participants with a strong view of life, feel a need for practicing beliefs, and believe that a spiritual force or power enables one to cope with personal events in life, have better psychological well-being, general positive affect, emotional and behavioral control, and less psychological distress, experience better emotional ties with others, less depression, more satisfaction in life and less anxiety.

# Canonical Correlation Function 1

Spiritual Beliefs and			Mental Health				
Life Att	Life Attitude Variables			ables			
	Set 1		Se	et 2			
Variable	Weight	Rank	Variable	Weight	Rank		
View of life	.455	1	Anxiety	396	8		
Practice	.371	2	Depression	495	6		
Day-to-day	.207		Loss	730	3		
Cope	.307	3	Affect	.784	2		
World affair	.101		Emotion	.560	5		
Natural disaster	.040		Satisfaction	.453	7		
			Distress	643	4		
			Well-being	.817	1		
			Optimism	.362			
			Pessimism	394			

#### Summary

This chapter dealt with the findings of the research that examined the relationship between spiritual belief, attitude toward life, and mental health. Two of the five null hypotheses had subhypotheses, and all were tested. A descriptive analysis of the sample was provided, including age, marital status, level of spiritual belief, distribution of religions, and spiritual understanding of life. Descriptive results of the survey were given, including the mean, standard deviation, and the minimum and maximum values of each spiritual belief measure (life satisfaction, depression, emotional ties, loss of behavioral/emotional control, anxiety, and general positive affect). Psychological distress and psychological well-being were additional subscales utilized in the study.

A direct-method multiple regression analysis was used to test hypotheses 1, 2, and 3. An independent t-test was run for hypothesis 4, and a canonical correlation analysis was used for hypothesis 5. Additionally, an inter-correlation matrix was run for hypothesis 2 in an effort to examine the relationship between the six predictor variables and the 10 criterion variables.

The research hypotheses analyses presented six significant relationships. The relationship between

spiritual belief and depression was significant. As view of life increased, depression significantly decreased in this group of participants. The results indicated that approximately 94% of the participants experienced depression only some of the time while 87% reported being happy most of the time. Furthermore, depression had a significant negative correlation with psychological well-being.

The relationship between spiritual beliefs and loss of behavioral/emotional control was significant. Participants with higher views of life revealed lower levels of uncontrollable behavior or emotion. A surprising 98% felt in control of their emotions most of the time. General positive affect, emotional ties, psychological distress, and psychological well-being also demonstrated significant relationships with spiritual belief.

The study revealed no relationship between spiritual belief and attitude toward life. However, view of life, a subcomponent used to measure spiritual beliefs, was significant. The results also indicated no relationships between spiritual beliefs and anxiety or life satisfaction.

### CHAPTER V

## SUMMARY, DISCUSSION, CONCLUSION, AND

## RECOMMENDATIONS

The final chapter presents a summary of the study, which includes the statement of the problem, purpose of the study, and an overview of the literature. The methodology utilized in the study, the analysis of the data results, and the conclusions and recommendations for further study are also presented.

### Summary of the Study

Religion and spirituality are important in the lives of most Americans. National polls show that 9 out of every 10 Americans believe in God and consider religion important in their lives (Elkins, 1999; Kroll, 1995; Kroll & Sheehan, 1989). It has become apparent that individuals are thinking about their spirituality or belief system, and may even be a bit curious as to the impact it has in their lives. More and more health professionals are also acknowledging that spirituality plays a relevant role, whether positively or negatively, in mental health (Elkins, 1999). In fact, over

85

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

the past decade, there has been an increased awareness of the impact of spirituality on mental health. Additionally, there has been growing evidence that attitude may also influence mental health (Seligman, 1998). Although it appears that attitude toward life and spiritual beliefs share common benefits for mental health, scant data have been gathered that look at this relationship. Studies on how one's spiritual beliefs contribute to one's life attitude and one's mental health are thus needed.

## Purpose of the Study

Therefore, this study investigated spirituality and its relationship with life attitude and mental health. It was the purpose of this study to look at a group of individuals and examine the relationship between spirituality, attitude toward life, and mental health.

The overall goal of this study was to provide a better understanding of spirituality and its relationship with mental health and inform health professionals of the role spiritual beliefs play in life attitude and mental health. Hopefully this will aid in the development of more effective strategies when working with clients seeking better mental health.

## Overview of the Literature

The field of psychology has produced very little research that supports the importance of spirituality, from a clinical stance (Mack, 1994). Some believe that spirituality is an area of study that often fuels more controversy than valid research (Elkins, 1999). However, the empirical research that has been conducted has repeatedly shown that a commitment to a spiritual relationship is usually associated with healthy physical, emotional, intellectual, occupational, and social functioning (Chandler et al., 1992; Ellison & Levin, 1998; Hawks, 1994; Maher & Hunt, 1993; Witmer & Sweeney, 1992). The unfortunate problem is the lack of an accurate, universal definition of spirituality, thus making the task of measurement of spirituality an enormous challenge. Some suggest this has had an impact on active contributors to research in the area of spirituality (Elkins, 1999).

Over the past several decades the interest in spirituality has appeared to increase, yet, reassurance in religion and religious leadership has continued to decrease. Studies show that although Americans may desire spirituality, it may not be in the form of organized religion (McGovern, 1998; Zinnbauer et al., 1997). Researchers have postulated several reasons why this may

exist:

"Spiritual health cannot be defined and researched and therefore, cannot be a legitimate area of study; spirituality is a part of the mental/emotional psychological dimension of health and does not need to be recognized as a separate entity within health education; and, recognition of spiritual health implies a recognition of religiosity which evades the issue of the separation of church and state in education" (Zinnbauer et al., 1997; p. 550).

Also supported by the literature is the view that spirituality seems to play a minimal role in the lives of most psychologists in the United States (Shafranske, 1996). A mere 33% of clinical psychologists have reported their faith as the most important influence in their lives, compared to 72% of the general population (Shafranske, 1996).

Several wellness models suggest that the spiritual dimension is the one component that is interactive with all the other dimensions (Adams & Bezner, 2000; Chandler et al., 1992; Crose et al., 1992; Goodloe & Arreola, 1992; Hettler, 1979; Witmer & Sweeney, 1992), indicating the importance of spirituality to mental health. It is believed, by some, that in order to experience wellness at an optimum level there must be a balance among the six dimensions of health (Adams & Bezner, 2000; Chandler et al., 1992; Hawks, 1994; Kolander & Chandler, 1990; Maher & Hunt, 1993; Witmer & Sweeney, 1992). Some have gone a step further and viewed

spirituality as not only an important component to the balance, but at the very core of wellness (Goodloe & Arreola, 1992; Kolander & Chandler, 1990; Witmer & Sweeney, 1992).

While some studies exist that explore the relationship of spirituality and mental health, little has been reported on the relationship of spiritual belief, attitude toward life, and mental health.

### Methodology

A sample of 690 individuals was obtained from the Northwest Athletic Club in Northwest, Indiana. Surveys were randomly distributed to club members, yielding a total of 187 (27% return rate) participants for the study. The data from these respondents were analyzed by zero-order correlation, multiple regression, or canonical correlation analyses, as seen appropriate.

## Findings and Discussion

Five research questions were addressed in this study. The analysis revealed that several questions resulted in significant relationships. Among the more important findings were the significant relationships found between spiritual beliefs, and the following: depression, loss of behavioral/emotional control, general positive affect, emotional ties, psychological distress, psychological wellbeing, and age.

Research question 1: Do people with high levels of spirituality have a better attitude toward life?

There was no statistically significant relationship found between attitude toward life, as measured by optimism and pessimism, and spiritual beliefs. This study did not support the studies that suggested that an optimistic outlook on life provides enrichment to life, both personally and professionally (Lee & Seligman, 1997; Peterson, 2000; Seligman, 1998). This group did not appear to have better attitudes toward life even though they reported feeling happy most of the time and experienced very little depression. Further, nearly 90% reported moderate to high levels of spirituality. A possible explanation for this finding could be that only those participants who felt secure with their spirituality responded to the survey, making them a fairly homogenous group. It should be noted that several non-participants (approximately 15%) indicated feeling hesitant about answering the spiritual experience instrument, stating this was personal information they were unwilling to share. Therefore, the opportunity for exploration of spiritual diversity may have been somewhat limited.

Research question 2: Do people with high levels of spirituality report better mental health?

There was some indication that spiritual beliefs affected mental health. Significant correlations were observed between spiritual belief and depression, loss of behavioral/emotional control, general positive affect, emotional ties, psychological distress, and psychological well-being. View of life emerged as the one consistent predictor variable in nearly every regression model that proved statistically significant. The findings in this study suggested that individuals with strong commitments to a particular spiritual belief system were generally: less depressed, more in control of their behavior or emotions, better able to establish emotional ties with others, and more psychologically fit. It is important to note the observation that many individuals use exercise as a means of reducing stress, tension, or anxiety in everyday life. It seems that those who regularly participate in exercise benefit from their efforts, and obtain a healthy body and mind.

It could also be speculated that "soft variables" may have played a part in the lack of significant correlations between spiritual beliefs and some mental health subscales. The literature refers to "soft variable" measures (i.e.,

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

paper-pencil personality tests) as measures that are typically limited by psychometric properties. Items such as self-actualization, rigidity, spirituality, or tolerance of ambiguity are only a few examples (Gartner et al., 1991). In contrast, "hard variables" are those variables that can be directly observed, reliably measured, value-neutral, and of unquestionable validity. Thus, falling under the category of soft variables, spirituality or spiritual belief is often difficult to measure.

Research question 3: Is age related to spiritual belief, attitude toward life, and mental health?

Spiritual beliefs, attitude toward life, and mental health, as a set, were significantly related to age. The strongest predictors included: anxiety, loss of emotional/behavioral control, and psychological distress. The older the subject, the more likely they were to experience higher levels of anxiety, higher levels of loss of emotional and behavioral control, and lower levels of psychological distress. This was a surprising finding because one might expect that with age comes a stronger, more balanced sense of self, encompassing spirituality. Another hypothesis might be that older participants were much less realistic in their thinking and more honest with themselves. Yet, for whatever reason, the older participants in this study reported moderate to high levels of spirituality, yet appeared more anxious and less in control of their emotions or behaviors.

Research question 4: Is there a difference in spiritual belief, attitude toward life, and mental health between men and women?

There were three variables that were statistically significant and were supported by the hypothesis: the belief that a spiritual power or force has an influence on day-to-day life, the belief that a spiritual power or force helps one cope with personal events in life, and psychological distress. Females presented as slightly more distressed than males and held slightly stronger beliefs in a spiritual power or force.

Given that the makeup of the participants resulted in a fairly homogeneous group, these findings were somewhat expected. The majority of the sample were happy most of the time, psychologically fit, possessed high general positive affect, and had a spiritual or religious understanding of their life. Had there been less homogeneity among individuals, a stronger difference between variables may have emerged. Clearly, the group as a whole presented as psychologically fit and spiritually sound, making it more

difficult to identify how the variables were related to each gender.

Research question 5: Is there a linear relationship between spiritual belief, attitude toward life, and mental health?

The first set of variates was statistically significant, revealing that a strong commitment to a particular belief system, practicing spiritual beliefs, and spiritual influences in personal events and in day-to-day life were associated with lower anxiety, depression, loss of behavioral and emotional control, psychological distress, and pessimism. These beliefs were also associated with higher general positive affect, emotional ties, life satisfaction, psychological well-being, and optimism.

Again, these findings suggested that individuals who present as psychologically and spiritually sound are generally representative of healthy physical, emotional, social, intellectual, and occupational functioning. This is a concept heavily supported by the current literature (Chandler et all., 1992; Ellison & Levin, 1998; Hawks, 1994; Maher & Hunt, 1993; Marquand, 1995; Witmer & Sweeney, 1992).

#### Conclusions

This research led to the following conclusions.

1. Although both male and female participants reported either a moderate or high level of spirituality, there appeared to be no significant correlation between spiritual belief and attitude toward life, as measured by optimism and pessimism.

2. There was some indication that spiritual beliefs affected mental health. This study revealed that participants with strong spiritual beliefs appeared psychologically sound. They presented with strong general positive affect and appeared to suffer significantly less psychological distress in their lives. Further, they appeared to be less depressed, more in control of their emotions and behaviors, and better able to establish emotional ties with others.

3. Spiritual belief, attitude toward life, and mental health, as a set, were significantly related to age. Specific predictors include anxiety, loss of emotional/behavioral control, and psychological distress. The older the participant, the more likely they were to experience high levels of anxiety and loss of behavioral/emotional control, and lower levels of psychological distress.

4. There appeared to be a linear relationship in spiritual belief, attitude toward life, and mental health between men and women. Females seemed to have a slightly greater belief that their day-to-day lives were influenced by a spiritual force or power, while embracing the idea that this force enabled them to better cope with personal events in life. Only psychological distress and psychological well-being differentiated the two groups, showing that females experienced higher psychological distress than males.

#### Implications

The following implications were a result of this research.

1. Although no significant relationship was found between spiritual beliefs and attitude toward life, this study suggested that the level of spiritual commitment might influence one's negative perception toward life. The more connected one was to a spiritual belief, the less negative they perceived life. Research which suggests that a negative attitude or negative perception of life impacts mental health (Peterson, 2000; Seligman & Csikszentmihalyi, 2000) seems to support this. As in this study, individuals who possess this negative attitude suffer with depression, low achievement, and bad health habits. Because census

reports suggest that the majority of Americans consider spirituality important in their lives, it is encouraged that clinicians find it relevant to investigate or explore the spiritual levels or the spiritual connections of their clients. Although many clinicians appear to remain hesitant to broach the subject with their clients, this study supports the literature that finds spirituality to be a key component to a balanced state of mental health.

2. Overall, the participants in this study held strong convictions for their spiritual beliefs while also appearing mentally sound. In some capacity, these participants' spiritual beliefs contributed to their strong mental health. Understanding what spirituality means to a client and how it impacts the social, intellectual, occupational, emotional, and physical dimensions of their lives might prove beneficial for clinicians. Assessing the level of commitment to and the meaning of one's spiritual belief system is essential information that should be included in the intake. Unfortunately, it is information often ignored by clinicians.

3. It appeared from this study that individuals with a strong commitment to a particular spiritual belief system generally experienced better mental health. Therefore, it could be implied that addressing issues in therapy with

these clients from a spiritual perspective might also foster better mental health. Clinicians can communicate to their clients their willingness to discuss issues of spirituality and provide a safe environment for that discussion without imposing their personal belief systems on their clients. Unfortunately, this is not a new concept. Twelve-step programs have recognized the importance of a relationship with a higher power and perceive it as an essential component to their program's foundation. Research continues to show that a higher power is key to optimum mental health (Chandler et al., 1992; Firshein, 1997; Hawks, 1994; Witmer & Sweeney, 1992). Spirituality has not received the full attention it deserves. This is evidenced by the hesitation of some clinicians to incorporate it into their work.

4. Because this study suggests the importance of strong correlations between spiritual beliefs and positive mental health, self-exploration of one's own level of belief might prove useful. For those uncertain about embracing spirituality into their own lives, there are a number of ways to still provide the client with the opportunity for spiritual expression. Educating one's self in the area of spirituality or exploring the subject with clients might aid clinicians in their efforts to understand the connection between spirituality and positive mental health. Addressing spiritual issues with clients is comparable to the work

clinicians already do with clients from different cultural backgrouds. Race, gender, sexual orientation, ethnic and cultural influences are all considered throughout the therapeutic process. One's religious or spiritual belief system warrants the same consideration.

#### Recommendations

1. Much of the literature in the area of spirituality is speculative and ungrounded in theory. The concept needs to be operationalized in order to better assist researchers with measurement in empirically based studies.

2. A replication of this study using a different population might yield even more successful results between spiritual beliefs, attitude toward life, and mental health. A less homogenous group could allow for more extensive exploration of spiritual beliefs.

3. In nearly every regression model, one individual predictor variable remained consistently significant. View of life was either the strongest predictor or the only predictor variable in most of the equations. The consistency of this variable warrants further exploration in future studies.

4. Further research needs to be conducted exploring the relationship between spirituality and religion. For instance, although Zinnbauer et al.'s (1997) and Mahoney and

Graci's (1999) studies both revealed that the majority of individuals considered themselves to be both spiritual and religious, the literature suggests that further investigation of these two concepts is warranted, as they are often used without definition and clarification of meaning.

5. For clinicians to begin to embrace the idea of spirituality in therapy, a paradigm shift in regard to spirituality and its validity to overall wellness is imperative. To foster this shift, more psychology programs might include in their curriculums training for spiritual issues. Students who receive this training might be more likely to see the importance of spirituality in people's lives and, thus, incorporate it into their therapeutic work. APPENDIX A

CORRESPONDENCE

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

# Andrews & University

September 7, 2000

Sharon Sacks 304 North Brook Dr. Michigan City, IN 46360

Dear Sharon:

RE: APPLICATION FOR APPROVAL OF RESEARCH INVOLVING HUMAN SUBJECTS								
HSRB Protocol # :	00-01 : 420	Application Type :	Original	Dept : Ed & Couns Psyc - 0104				
Review Category :	Exempt	Action Taken :	Approved					
Protocol Title :	The Relationship B Filness Participant	etween Spiritual Beli s in Northern Indiani	ef, Life Attitude 1	and Mental Health Among Physical				

On behalf of the Human Subjects Review Board (HSRB) I want to advise you that your proposal has been reviewed and approved. You have been given clearance to proceed with your research plans.

All changes made to the study design and/or consent form after initiation of the project require prior approval from the HSRB before such changes are implemented. Feel free to contact our office if you have any questions.

The duration of the present approval is for one year. If your research is going to take more than one year, you must apply for an extension of your approval in order to be authorized to continue with this project.

Some proposal and research designs may be of such a nature that participation in the project may involve certain risks to human subjects. If your project is one of this nature and in the implementation of your project an incidence occurs which results in a research-related adverse reaction and/or physical injury, such an occurance must be reported immediately in writing to the Human Subjects Review Board. Any project-related physical injury must also be reported immediately to the University physician, Dr. Loren Hamel, by calling (616) 473-2222.

We wish you success as you implement the research project as outlined in the approved protocol.

Sincerely.

Linda Thorman, Ed.D. Human Subjects Review Board c: Nancy Carbonell

> Office of Scholary Research, Graduate Dean's Office, (616) 471-6361 Andrews University, Berlien Springs, MI, 49104-0640

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

# Andrews University

Greetings! My name is Sharon Sacks and I am a doctoral student at Andrews University pursuing a degree in Psychology. As a future psychologist I am interested in studying what contributes to a healthy outlook on life. What can prevent depression or anxiety attacks before they happen? Several studies have been conducted in this area, yet few, if any, have studied those individuals who have made a commitment to good physical health. This is why I am inviting you to participate in this study.

As a thank you for your time and effort I am offering to the first <u>200</u> respondents a *chance* to **win three months membership FREE!** All you need to do is complete the enclosed survey and turn it into the front desk at the Northwest Athletic Club by **October 2nd**. Keep the attached ticket at the bottom of this page and on *October 4th* **two numbers will be drawn** and posted at the NAC. The lucky individuals who hold the matching ticket numbers will *each* receive three months of their membership paid. *Remember, the matching ticket number <u>must</u> be presented in order to receive the three month membership.* I appreciate your time in completing this survey.

Sincerely,

Sharon Sacks, M.S.

Ph.D. Candidate

Place this ticket on your refrigerator for safe keeping.

### Andrews University

Greetings Northwest Athletic Club members:

Approximately two weeks ago you received a survey by mail along with a cover letter explaining my research project. Many of you returned the survey, and I thank you. For those of you who have not yet returned your survey, I ask that you take 10 minutes to fill out the information and return it to either the NAC or directly to me.

Literature shows that spirituality has been an often overlooked dimension of mental health and may actually be an important component of optimum wellness. Research that explores spirituality and its effects on one's attitude toward life and mental health is scarce and sorely needed. My hope is that I will be able to use the results of this study to more fully understand the impact of spirituality on mental health. Your willingness to participate will make this study possible.

I can assure you that confidentiality will be ensured throughout handling the survey. You are <u>not</u> asked to place your name or any other personal information that will identify the survey with you as a person. The information will be used only for statistical calculations.

If you have any questions about the survey, please fell free to contact me at the phone number or the address below. I do appreciate your time for completing the questionnaire.

Sharon Sacks:

	Phone: 219-879-6797						
	Address: 304 Northbrook Drive						
	Michigan City, IN 46360						
Advisor:	Dr. Nancy Carbonell						
	Andrews University						
	Educational and Counseling Psychology						
	Bell Hall						
	Berrien Springs, MI 49104						

APPENDIX B

INSTRUMENTS

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

### The Royal Free Questionnaire on Beliefs and Experiences

#### This questionnaire concerns your beliefs and views about life. To start, could you tell us your:

Age	Gender: Male	Female	Fitn	ess level:		0 - 4 hrs. per week
					þ	5 - 8 hrs. per week
					С	9+ hrs. per week
How would you de	escribe yourself ( 🖌	one or more)				
Married	Living with a	partner	Divorced Widow	ved Sepa	rated	Single
How would you de	escribe yourself ( 🖌	one)?				
Caucasia	an African A	merican	Asian Hispani	c Other		
Are you?						
Employed Une	employed seeking v	vork Stud	ent <u>Retired</u> H	ome manager	0	n sick leave
If you are employed	d, can you describe	your work?	······			
If you are not empl	oyed now, what wa	as your last job	o??			

## You will now be asked some questions about your religious and spiritual beliefs. Please try to answer them even if you have little interest in religion.

In using the word *religion*, we mean the actual practice of a faith, e.g. going to a temple, mosque, church or synagogue. Some people do not follow a specific religion but do have *spiritual* beliefs or experiences. For example, they may believe that there is some power or force other than themselves that might influence their life. Some people think of this as God or gods, others do not. Some people make sense of their lives without any religious or spiritual belief.

1. Therefore, would you say that you have a *religious* or *spiritual* understanding of your life? (Please  $\checkmark$  one or more).

\_\_\_\_ Religious

\_\_\_\_ Spiritual

Neither religious or spiritual

#### If you have NEVER had a RELIGIOUS or SPIRITUAL BELIEF, please go to Question 13 on page 3.

#### Otherwise, PLEASE TRY TO ANSWER THE FOLLOWING QUESTIONS:

2. Can you explain briefly what form your religious spiritual belief has taken?

3.	Some people hold strongly to their views and oth religious spiritual view of life? Circle the number			•	to your	
	0 1 2 3 4	5 6	7	8	9	10
	Weakly held view			Strongly	held vie	ew.
4.	Do you have a specific religion?					
	I do not observe a religion (go to question 8 Methodist Judaism Baptist E		tholic	_ Presbyter	ian (	Other
5. 6.	Can you give more detail? (e.g. denomination, se Do any of the following play a part in your belief		ou might n	ray or med	utate alon	e or with
	other people. ( $\checkmark$ as many choices as apply to y		sa migin p	luy of filed	ate atom	
	Prayer	Al	one	Wit	h other p	cople
	Ceremony (e.g. washing before prayer,) a religious service	AI	one	Witi	h other p	eople
	Meditation	Al	one	Witl	n other p	eople
	Reading and study	Ab	one	With	n other p	eople
	Contact with religious leader	Ale	one	With	n other p	cople
	None of the above					
7.	How important to you is the practice of your belied day life? Please circle the number on the scale wh			-	ices) in j	your day-to-
	0 1 2 3 4 Not Essentia)	5 6	7	8	9 Necessa	10
You ca	an explain further if you would like to:					ı y
8.	Do you believe in a spiritual power or force other day-to-day life? Please circle the number on the se				appens to	) you in our

0	1	2	3	4	5	6	7	8	9	10
No influ	ience								Strong in	nfluence

					108					
9.	Do you believe events in your l									nally with
	0 l No Help	2	3	4	5	6	7	8	9 A gr	10 eat help
10.	Do you believe Please circle the							ces world	affairs, e	.g. wars?
	0 1 No influence	2	3	4	5	6	7	8	9 Stroi	10 ng influenc
11.	Do you believe earthquakes, flo									s, like
	0 1 No influence	2	3	4	5	6	7	8	9 Stror	10 ng influence
2.	Do you commu via a medium?	nicate in a	ny way w	ith a spirit	ual power	, for exam	ple by pra	yer or con	tact	
	Yes/ N	No: Unsur	e lfyes	. describe	form of co	ommunica	tion.			
3.	Yes/ M	<del>.</del>					tion.			
13.	Do you think th	<del>.</del>	t in some		our death		tion.			
	Do you think th	at we exist No Unsure lad an <i>inte</i> felt at one	t in some e If yes, <i>use exper</i> .	form after . describe f <i>tence</i> (unro world or u	our death he form: clated to d niverse. (1	rugs or al	cohol) in v eve in Go	d it may h		
	Do you think the Yes- M Have you ever h meaning in life, experience of G	at we exist No Unsure lad an <i>inte</i> felt at one	t in some e If yes. <i>mse exper</i> - e with the ght have b	form after . describe f <i>tence</i> (unro world or u	our death he form: clated to d niverse. (1	rugs or al	cohol) in v eve in Go	d it may h		
14.	Do you think the Yes- M Have you ever h meaning in life, experience of G	at we exist No. Unsure felt at one od.) It mig No: Unsure	t in some e If yes. <i>nse exper</i> . e with the ght have b	form after , describe f <i>ience</i> (unre world or u been for a f	our death he form: elated to d niverse. () Tew mome	rugs or al- f you beli nts, hours	cohol) in v eve in Go or even d	d it may h; ays.	ave felt li	
13. 14. 1 <b>f you</b> 15.	Do you think the Yese Y Have you ever h meaning in life, experience of Ge Yese N	at we exist No Unsure lad an <i>inte</i> felt at one od.) It mig No: Unsure <b>his questic</b>	t in some e If yes. <i>mse exper</i> e with the ght have b e <b>on. go on</b>	form after , describe f <i>tence</i> (unre world or u been for a f <b>to questio</b>	our death he form: elated to d niverse. (1 Tew mome n 18. If y	rugs or al f you beli nts, hours es or unst	cohol) in v eve in Go or even d	d it may h; ays.	ave felt li	

- 17. Can you describe it?
- 18. Some people have described intense experiences at a time when they almost died but were eventually revived. Has this ever happened to you?

Yes: No. Unsure - If yes or unsure, please describe the experience

If YES or UNSURE, how much has this near death experience changed your life? Please circle the number on the scale which best describes your view.

0	1	2	3	4	5	6	7	8	9	10

Not at all

Extremely

#### 110

### Mental Health Inventory

These next questions are about how you feel, and how things have been with you mostly within the past month.

For each question, please circle a number for the <u>one answer</u> that comes <u>closest</u> to the way you have been feeling.

54	How happy, satisfie	d, or ple	eased have	e you been wit	th you	r personal lite d	uring th	
	past month?							
	(Circle one)							
	Extremely happy		1	Sometimes fai	rly satis	sfied or unhappy	4	
	Very happy most of t	he time	2	Generally diss	tisfied	, unhappy	5	
	Generally satisfied, please		3	Very dissatisfied, unhappy most of the tim				
55	How much of the ti	me have	e you felt l	onely during	the pa	st month?		
	All the time	1	A good	bit of the time	3	A little of the	time 5	
	Most of the time	2	Some of	the time	4	None of the ti	me 6	
56	How often did you l unexpected situations during th			r jumpy when	faced	with excitemen	t or	
56	unexpected				faced	with excitemen Almost never	t or 5	
56	unexpected situations during th	e past m	ionth?	ten				
56 57	unexpected situations during th Always	e past m l 2	ionth? Fairly of Sometin	ten ies	3 <del>1</del>	Almost never Never	5 6	
	unexpected situations during th Always Very often During the past mor hopeful	e past m l 2	ionth? Fairly of Sometin	ten ies the time have	3 <del>1</del>	Almost never Never	5 6	
	unexpected situations during th Always Very often During the past mon hopeful and promising?	e past m 1 2 nth, how	oonth? Fairly of Sometin v much of	ten 1es the time have ten	3 4 e you f	Almost never Never elt that the futu	5 6 Ire looks	
	unexpected situations during th Always Very often During the past mor hopeful and promising? Always	e past m 1 2 nth, how 1 2	Fairly of Sometin w much of Fairly of Sometin	ten 1es the time have ten	3 4 2 you f	Almost never Never elt that the futu Almost never	5 6 ire looks 5	
57	unexpected situations during th Always Very often During the past mor hopeful and promising? Always Very often	e past m 1 2 nth, how 1 2	Fairly of Sometin w much of Fairly of Sometin	ten the time have ten tes	3 4 2 you f	Almost never Never elt that the futu Almost never	5 6 ire looks 5	

			111		
59	How much of the time were interesting to you		the past month, has you	ır daily	life been full of things that
	All the time	1	A good bit of the time	3	A little of the time 5
	Most of the time	2	Some of the time	4	None of the time 6
60	How much of the tir tension?	ne, duri	ng the past month, did	l you fe	eel relaxed and free of
	All the time	1	A good bit of the time	3	A little of the time 5
	Most of the time	2	Some of the time	4	None of the time 6
61	During the past mon things you do?	ith, how	much of the time hav	e you g	generally enjoyed the
	All the time	1	A good bit of the time	3	A little of the time 5
	Most of the time	2	Some of the time	4	None of the time 6
62		control	e you had any reason to over the way you act, it not enough to		
	Maybe a little 2	be con	cerned or worry about it ad I have been a little	5	concerned
		concer		4	Yes, and I am very 6 much concerned about it
53	In general, would you Yes, definitely	u say yo l	ur morals have been al I don't know	ove re 3	proach? Definitely not 5
	Yes, probably	2	Probably not	4	
64	Did you feel depresse	d durin	g the past month?		
	Yes, to the point I did not care about it anything for days at a t	l .ime	Yes, quite depressed alm everyday	nost 3	No, never felt depressed at all 5
	Yes, very depressed almost every day	2	Yes, a little depressed n and then	ow 4	

65	During the past mor	nth, how	v much of the time hav	re you fe	lt loved and wa	nted?
	All the time	1	A good bit of the time	3	A little of the ti	ime 5
	Most of the time	2	Some of the time	4	None of the tin	1e 6
66	How much of the tir person?	ne, duri	ing the past month, ha	ve you h	een a very nerv	ous
	All the time	1	A good bit of the time	3	A little of the ti	me 5
	Most of the time	2	Some of the time	4	None of the tin	ne 6
67	When you got up in expect to have an int		rning, this past month, g day?	about l	10w often did y	ou
	Always	1	Fairly often	3	Almost never	5
	Very often	2	Sometimes	4	Never	6
68	How often have ther coward?	e ben ti	mes in your life when y	vou felt	you acted like a	
	Very often	1	Sometimes	3	Never	5
	Fairly often	2	Almost never	4		
69	During the past mon strung"?	th, how	much of the time have	e you fe	lt tense or "higl	1-
	All the time	1	A good bit of the time	3	A little of the ti	me 5
	Most of the time	2	Some of the time	4	None of the tim	e 6
70	During the past mon thoughts, emotions, t		e you been in firm cont ?	rol of y	our behavior,	
	Yes, very definitely	1	Yes, I guess so 3	No, and disturb	l I am somewhat ed	5
	Yes, for the most part	2	No, not to well 4		l I am very	6
71	During the past mon something?	th, how	often did your hands s			do
	Always	1	Fairly often 3	Almost	never	5
	Very often	2	Sometimes 4	Never		6

72	During the past mor forward to?	nth, hov	w often did you feel tha	it you h	ad nothing to lo	ook					
	Always	1	Fairly often	3	Almost never	5					
	Very often	2	Sometimes	4	Never	6					
73	Would you say that	you give	e every penny you can	to chari	ty?						
	Yes, definitely	1	Yes, I try	3							
	Yes, for the most part	2	No	4							
74	How much of the time, during the past month, have you felt calm and peaceful?										
	Yes, definitely	1	Yes, I try	3							
	Yes, for the most part	2	No	4							
75	How much of the time, during the past month, have you felt emotionally stable?										
	All the time	1	A good hit of the time	3	A little of the ti	ime 5					
	Most of the time	2	Some of the time	4	None of the tin	ne 6					
76	blue?		ng the past month, hav	·							
	All the time	1	A good bit of the time	3	A little of the ti						
	Most of the time	2	Some of the time	4	None of the tim	ne 6					
77	How often have you	felt like	crying, during the pas	t montl	h?						
	Always	1	Fairly often	3	Almost never	5					
	Very often	2	Sometimes	4	Never	6					
78	In choosing your frie religion, or their poli		w important to you are liefs?	e things	like their race,	their					
	Always very important	1	Usually important 3	Hardly	ever important	5					
	Almost always importan	nt 2	Not too important 4	Not im	portant at all	6					
79	During the past monty you were dead?	th, how	often did you feel that	others	would be better	off if					
	Always	1	Fairly often	3	Almost never	5					
	Very often	2	Sometimes	4	Never	6					

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

80	How much of the tir	ne, dur	ing the past month, we	re you a	able to relax with	out
	difficulty?					
	All the time	1	A good bit of the time	3	A little of the tin	ne 5
	Most of the time	2	Some of the time	4	None of the time	6
81			wmuch of the time did ing loved, were full and		-	
	All the time	1	A good bit of the time	3	A little of the tim	ie 5
	Most of the time	2	Some of the time	4	None of the time	6
82	How often, during th the way you wanted i		month, did you feel tha	ıt nothi	ng turned out for	. you
	Always	1	Fairly often	3	Almost never	5
	Very often	2	Sometimes	4	Never	6
83	How much have you past month?	been ho	othered by nervousness	, or you	ır "nerves," durin	g the
	Extremely so, to the point where I could not take care of things	1	Bothered quite a bit by nerves	3	Bothered just a little by nerves	5
	Very much bothered	2	Bothered some, enough to notice	4	Not bothered at all by this	6
84	During the past mont adventure for you?	th, bow	much of the time has	living b	een a wonderful	
	All the time	1	A good bit of the time	3	A little of the time	e 5
	Most of the time	2	Some of the time	4	None of the time	6
85	If it is more convenie	nt for y	ou to do so, how often	will yo	u tell a lie?	
	Very often tell a lie	1	Sometimes tell a lie	3	Never tell a lie 5	
	$\mathbf{E} \cdot \mathbf{I} = \mathbf{C}$	2	41 .			

86	How often, during nothing could cheer		month, have you felt so ?	o down	own in the dumps that		
	Always	1	Fairly often	3	Almost never	5	
	Very often	2	Sometimes	4	Never	6	
87	During the past mo	nth, did	you ever think about t	aking y	your own life?		
	Yes, very often	1	Yes, a couple of times	3	No, never	5	
	Yes, fairly often	2	Yes, at one time	4			
88	During the past month, how much of the time have you felt restless, fidgety, or impatient					ety, or	
	All the time	1	A good bit of the time	3	A little of the t	ime 5	
	Most of the time	2	Some of the time	4	None of the ti	ne 6	
89	How often have you done anything of a sexual nature that society does not approve of?					ot	
	Very often	1	Sometimes	3	Never	5	
	Fairly often	2	Almost never	4			
90	During the past mo about things?	nth, hov	v much of the time have	e you b	ı ben moody or brooded		
	All the time	1	A good bit of the time	3	A little of the t	ime 5	
	Most of the time	2	Some of the time	4	None of the tir	ne 6	
91	How much of the time, during the past month, have you felt cheerful, light- hearted?					ht-	
	All the time	1	A good bit of the time	3	A little of the t	ime 5	
	Most of the time	2	Some of the time	4	None of the tin	ne 6	
92	During the past mor Always	nth, how 1	often did you get rattle Fairly often	ed, ups 3	et, or flustered? Almost never	5	
	Very often	2	Sometimes	4	Never	6	
	-						

93	Are your table manners at home just as good as they are when you are inv to dinner?						
	Yes, always just as good	11	Yes, usually just as goo	od 3	No, quite a bit 5 worse at home		
	Yes, with rare exception	n 2	No, usually worse at he	ome 4	No, very bad at home 6		
94	During the past month, have you been anxious or worried?						
	Yes, extremely so, to the point of being sick or almost sick	1	Yes, quite a bit	3	A little of the time 5		
	Yes, very much so	2	Yes, some, enough to bother me	4	None of the time 6		
95	During the past mont	th, how	much of the time wer	ere you a happy person?			
	All the time	1	A good bit of the time	3	A little of the time 5		
	Most of the time	2	Some of the time	4	None of the time 6		
96	6 How often during the past month did you find yourself having difficulty tr calm down?						
	Always	1	Fairly often	3	Almost never 5		
	Very often	2	Sometimes	4	Never 6		
97	During the past mont spirits?	h, how	much of the time have	e you he	en in low or very low		
	All the time	1	A good bit of the time	3	A little of the time 5		
	Most of the time	2	Some of the time	4	None of the time 0		
98	How often, during the rested?	e past n	nonth, have you been v	vaking	up feeling fresh and		
	Always, every day	1	Most days	3	Hardly ever 5		
	Almost, every day	2	Some days, but usually not	4	Never wake up 6 feeling rested		

99 During the past month, have you been under or felt you were under any strain, stress, or pressure?

Yes, almost more than I could stand or bear	1	Yes, some, more than usual	3	Yes, a little bit	õ
Yes, quite a bit of	2	Yes, some, but about	4	No, not at all	6
pressure		normal			

Instructions: The 56 statements printed below represent individual differences in viewpoint. Using the scale shown below, please respond with your own point of view to all of the statements: for example, if you strongly agree with a statement then circle 1 (SA). Do not spend a lot of time thinking about each one; just indicate your first impression. Remember, respond to theses statements according to how you feel about them right now.

1- strongly agree 2- agree 3- disagree 4- strongly disagree

	SA	А	D	SD
1. I like people I get to know.	1	2	3	4
2. It is best not to set your hopes too high since you will probably				
be disappointed.	1	2	3	4
3. There is so much to be done and so little time to do it in.	1	2	3	4
4. I have a tendency to make mountains out of molehills.	1	2	3	4
5. Rarely do I expect good things to happen.	1	2	3	4
6. Everything changes so quickly these days that I often have trouble				
deciding which are the right rules to follow.	1	2	3	4
7. All in all the world is a good place.	1	2	3	4
8. When it comes to my future plans an ambitions in life, I expect				
more to go wrong than right.	1	2	3	4
9. My hardest battles are with myself.	1	2	3	4
10. I believe there's not much hope for the human race.	1	2	3	4
11. It does not take long t shake off a bad mood.	1	2	3	4
12. If you hope and wish for something long and hard enough, you will				
eventually get it.	1	2	3	4

13. People get ahead by using 'pull' and not because of what they know.	1	2	3	4
14. Even when things in my life are going okay, I expect them to get				
worse soon.	1	2	3	4
15. With enough faith, you can do almost anything.	1	2	3	4
16. I enjoy myself most when I am alone, away from other people.	1	2	3	4
17. When I undertake something new, I expect to succeed.	1	2	3	4
18. Honesty is the best policy in all cases.	1	2	3	4
19. I generally look at the brighter side of life.	1	2	3	4
20. If I make a decision on my own, I can pretty much count on the fact				
that it will turn out to be a poor one.	1	2	3	4
21. I generally make light of my problems.	1	2	3	4
22. It is always a good thing to be frank.	1	2	3	4
23. Where there's a will, there's a way.	1	2	3	4
24. I have a tendency to blow up problems so they seem worse than				
they really are.	1	2	3	4
25. All in all, it is better to be humble and honest than important and				
dishonest.	1	2	3	4
26. As time goes on, things will most likely get worse.	1	2	3	4
27. It is the slow, steady worker who usually accomplishes the most				
in the end.	1	2	3	4
28. When I go to a party I expect to have fun.	ł	2	3	4
29. Times are getting better.	1	2	3	4
30. Everyone should have an equal chance and an equal say.	1	2	3	4
31. Better to expect defeat: then it doesn't hit so hard when it comes.	1	2	3	4
32. It is wise to flatter important people.	1	2	3	4

33. I expect to achieve most of the things I want to in life.	1	2	3	4
34. It seems the cards of life are stacked against me.	1	2	3	4
35. What is lacking in the world today is the old kind of friendship				
that lasted for a lifetime.	1	2	3	4
36. When the weatherman predicts 50% chance rain, you might just as				
well count on seeing rain.	1	2	3	4
37. Before an interview, I am usually confident that things will go well.	1	2	3	4
38. Sometimes I feel down, but I bounce right back again.	1	2	3	4
39. The future seems too uncertain for people to make serious plans.	1	2	3	4
40. When I have undertaken a task, I find it difficult to set it aside even				
for a short time.	1	2	3	4
41. Tenderness is more important than love.	1	2	3	4
42. When gambling, I expect to lose.	1	2	3	4
43. Anybody who is willing to work hard has a good chance for				
succeeding	1	2	3	4
44. The future looks very dismal.	1	2	3	4
45. If I had to choose between happiness and greatness, I'd choose				
greatness.	1	2	3	4
40. Minor setbacks are something I usually ignore.	1	2	3	4
47. In general, things turn out all right in the end.	1	2	3	4
48. It is better to be a dead hero than a live coward.	1	2	3	4
49. Give me 50/50 odds and I will choose the wrong answer every time.	1	2	3	4
50. It is hard to get ahead without cutting corners here and there.	1	2	3	4
51. If I were in competition and contestants were narrowed down to				
myself and one other person, I would expect to be runner-up.	1	2	3	4

52. April showers bring may flowers.	1	2	3	4
53. I can be comfortable with nearly all kinds of people	1	2	3	4
54. The worst defeats come after the best victories.	1	2	3	4
55. In the history of the human race there have prohably been just a				
handful of really great thinkers.	1	2	3	4
56. Every cloud has a silver lining.	1	2	3	4

REFERENCE LIST

Reproduced with permission of the copyright owner. Further reproduction prohibited without permission.

#### REFERENCE LIST

- Adams, T., & Bezner, J. (2000). Conceptualization and measurement of the spiritual and psychological dimensions of wellness in a college population. Journal of American College Health, 48(4), 165-173.
- Aldridge, D. (1991). Spirituality, healing and medicine. British Journal of General Practice, 41, 425-427.
- American Psychological Association. (1994). Diagnostic and statistical manual of mental disorders (4<sup>th</sup> ed.). Washington, DC: Author.
- American Psychological Association. Special issue on happiness, excellence, and optimal human functioning. American Psychologist, 55(1), 5-185.
- Bensley, R. J. (1991). Spiritual health as a component of worksite health promotion/wellness programming: A review of the literature. Journal of Health Education, 22(6), 352-353.
- Bergin, A. E. (1983). Religiosity and mental health: A critical re-evaluation and meta-analysis. Professional Psychology: Research and Practice, 14, 170-184.
- Bergin, A. E., & Jensen, J. P. (1990). Religiosity of psychotherapists: A national survey. Psychotherapy, 27, 3-7.
- Bilgrave, D., & Deluty, R. (1998). Religious beliefs and therapeutic orientations of clinical and counseling psychologists. Journal for the Scientific Study of Religion, 37(2), 329-349.
- Chandler, C. K., Miner-Holden, J., & Kolander, C. A. (1992). Counseling for spiritual wellness: Theory and practice. Journal of Counseling and Development, 71, 168-175.

- Chapman, L. (1986). Spiritual health: A component missing from health promotion. American Journal of Health Promotion, 1(1), 38-41.
- Chirban, J. T. (1992). Healing and spirituality. Pastoral Psychology, 40(4), 235-245.
- Crose, R., Nicholas, D. R., Gobble, D. C., & Frank, B. (1992). Gender and Wellness: A multidimensional systems model for counseling. Journal of Counseling & Development, 71(2), 149-156.
- Dember, W., & Brooks, J. (1989). A new instrument for measuring optimism and pessimism: test-retest reliability and relations with happiness and religious commitment. Bulletin of the Psychonomic Society, 27(4), 365-366.
- Dember, W. N., Martin, S. H., Hummer, M. K., Howe, S. R., & Melton, R.S. (1989). The measurement of optimism and pessimism. Current Psychology: Research and Reviews, 8(2), 102-119.
- Elkins, D. N. (1999). Spirituality. Psychology Today, 32(5), 44-48.
- Ellison, C., & Levin, J. (1998). The religion-health connection: Evidence, theory, and future directions. Health Education & Behavior, 25(6), 700-720.
- Ellison, C., & Smith, J. (1991). Toward an integrative measure of health and well-being. *Journal of Psychology and Theology*, 19(1), 35-43.
- Fahlberg, L. L., & Fahlberg, L. A. (1991). Exploring spirituality and consciousness with an expanded science: Beyond the ego with empiricism, phenomenology, and contemplation. American Journal of Health Promotions, 5(4),273-281.
- Firshein, J. (1997). Spirituality in medicine gains support in the USA. Lancet, 349(9061), 1300-1302.
- Gallup, G. H. (1993). Religion in America: 1992-1993. Princeton, NJ: Gallup Organization.

Gallup, G. H. (1995). The Gallup Poll: Public 1995. Wilmington, DE: Scholarly Resources.

- Gartner, J., Larson, D. B., & Allen, G.D. (1991). Religious commitment and mental health: A review of the empirical literature. Journal of Psychology and Theology, 19(1), 6-25.
- George, L. K., Larson, D. B., Koenig, H. G., & McCullough, M. E. (2000). Spirituality and health: What we know, what we need to know. Journal of Social and Clinical Psychology, 19(1), 102-116.
- Goodloe, N., & Arreola, P. (1992). Spiritual health: Out of the closet. Journal of Health Education, 23(4), 221-226.
- Hamilton, D., & Jackson, M. (1998). Spiritual development: Paths and processes. Journal of Instructional Psychology, 25(4), 262-270.
- Hawks, S. (1994, Summer). Spiritual health: Definition and theory. Wellness Perspectives, 10(4), 3-13.
- Hettler, W. (1979). Six dimensions of wellness. Stevens Point, WI: National Wellness Institute, University of Wisconsin.
- Hill, P., & Hood, R. (1999). Measures of Religiosity. Birmingham, AL.: Religious Education Press.
- Hill, P. C., Pargament, K. I., Hood, Jr., R. W., McCullough, M. E., Swyers, J. P., Larson, D. B., & Zinnbauer, B. J. (2000). Conceptualizing religion and spirituality: Points of commonality, points of departure. Journal for the Theory of Social Behaviour, 30(1), 51-77.
- Hood, R., Spilka, B., Hunsberger, B., & Gorsuch, R (1996). The psychology of religion: An empirical approach. New York, NY: Guilford Press.
- Hoyman, H. (1962). Our modern concept of health. Journal of School Health, 32(9), 253-263.
- Hummer, M., & Dember, W. (1992). On partial independence of optimism and pessimism. Current Psychology, 11(1), 30-37.

- Jose, N. (1987). The silent gift: A project for spiritual health. Journal of School Health, 57(2), 72-73.
- Joseph, M. (1998). The effect of strong religious beliefs on coping with stress. Stress Medicine, 14, 219-224.
- King, M., & Dein, S. (1998). The spiritual variable in psychiatric research. Psychological Medicine, 28(6), 1259-1262.
- Koenig, H. G. (1994). Religion and hope for the disabled elder. In J.S. Levin (Ed.), Religion in aging and health: Theoretical foundations and methodological frontiers. Thousand Oaks, CA: Sage.
- Koenig, H. G. (1995). Research on religion and aging: An annotated bibliography. Westport, CT: Greenwood.
- Koenig, H. G., George, L. K., & Peterson, B. L. (1998). Religiosity and remission of depression in mentally ill older patients. American Journal of Psychiatry, 155(4), 536-542.
- Kolander, C., & Chandler, C. (1990, April). Spiritual health: A balance of all dimensions. Paper presented at the Annual Meeting of the American Alliance for Health, Physical Education, Recreation and Dance, New Orleans, LA.
- Kroll, J., (1995). Religion and psychiatry. Current Opinion in Psychiatry, 8, 335-339.
- Kroll, J. (1998). Religion and psychiatry. Current Opinion in Psychology, 11(5), 559-562.
- Kroll, J. & Sheehan, W. (1989). Religius beliefs and practices among 52 psychiatric inpatients in Minnesota. American Journal of Psychiatry, 146(1), 67-72.
- Larson, D. B., Larson, S. L., & Puchalski, C. M. (2000, January). The once-forgotten factor in psychiatry, part1: Residency training addresses religious and spiritual issues. Psychiatric Times, 18-23.

- Larson, D. B., Pattison, E. M., Blazer, D. G., Omran, A. R., & Kaplan, B. H. (1986). Systematic analysis of research on religious variables in four major psychiatric journals, 1978-1982. American Journal of Psychiatry, 143, 329-334.
- Lee, Y., & Seligman, M. (1997). Are Americans more optimistic than the Chinese? *Personality and Social Pyschology Bulletin, 23, 32-40.*
- Lewis, L., & Dember, W. (1995). Can experimentally induced mood effect optimism and pessimism scores? Current Psychology, 14, 29-42.
- Mack, M. L. (1994). Understanding spirituality in counseling psychology: Considerations for research, training, and practice. Counseling and Values, 39(1), 15-31.
- Maher, M., & Hunt, T. (1993). Spirituality reconsidered. Counseling and Values, 38, 21-28.
- Mahoney, M., & Graci, G. (1999). The meaning and correlates
   of spirituality: Suggestions from an exploratory survey
   of experts. Death Studies, 23(6), 521-528.
- Marinelli, R., & Plummer, O. (1999). Healthy aging: Beyond exercise. Activities, Adaptation & Aging, 23(4), 1-11.
- Marquand, R. (1995). Healing role of spirituality gains ground. [On-line]. Christian Science Monitor, 88(8), 1. Abstract retreived December 6, 1995, from EBSCO Host file.
- McConnell, A., & Bill, C. (1993). Personality through metaphor: Optimism, pessimism, locus of control, and sensation seeking. Current Psychology, 12, 195-216.
- McGovern, C. (1998, August 24). Good for the soul, good for the mind. Alberta Report, 25, 22-25.
- McIntosh, D. N., Silver, R. C., & Wortman, C. B. (1993). Religion's role in adjustment to a negative life event: coping with the loss of a child. Journal of Personality and Social Psychology, 65, 812-821.

- McKee, D. D., & Chappel, J. N. (1992). Spirituality and medical practice. Journal of Family Practice, 35(2), 205-208.
- Myers, D. G. (2000). The funds, friends, and faith of happy people. American Psychologist, 55(1), 56-67.
- Myers, J. E. (1992). Wellness, prevention, development: the cornerstone of the profession. *Journal of Counseling* and Development, 71, 136-139.
- Osman, J., & Russell, R. (1979). The spiritual aspects of health. Journal of School Health, 49, 359.
- Pargament, K. I. (1999). The psychology of religion and spirituality? Yes and no. The International Journal for the psychology of religion, 9(1), 3-16.
- Peterson, C. (2000). The future of optimism. American Psychologist, 55(1), 44-55.
- Reichle, P., Vantrease, J., & Fink, E. (1989, April). Spiritual health: The next frontier. Paper presented at the meeting of the Association for the Advancement of Health Education, Boston, MA.
- Reilley, S., & Dember, W. (1998). Optimism, pessimism, and friendship. Current Psychology, 17, 3-20.
- Richards, P., & Bergin, A. (1997). A spiritual strategy for counseling and psychotherapy. Washington, DC: American Psychological Association.
- Richardson, G. E., & Nolan, M. P. (1984). Treating the spiritual dimension through visual imagery. *Health Values*, 8(6), 25-30.
- Roof, W. C. (1993). A generation of seekers: The spiritual journeys of the baby boom generation. San Francisco: Harper.
- Salovey, P., Rothman, A. J., Detweiler, J. B., & Steward, W. T. (2000). Emotional states and physical health. American Psychologist, 55(1), 110-121.

- Shafranske, E. P. (1996). Religion and the clinical practice of psychology. Washington, DC: American Psychological Association.
- Shafranske, E., & Maloney, N. (1990). California
   psychologists' religiosity and psychotherapy. Journal
   of Religion and Health, 29, 219-231.
- Seaward, B. L. (1991). Spiritual well-being: A heal education model. Journal of Health Education, 22, 166-169.
- Seligman, M. E. (1998). Learned optimism: How to change your mind and your life. New York, NY: Simon and Schuster Inc.
- Seligman, M., & Csikszentmihalyi, M. (2000). Positive Psychology. American Psychologist, 55(1), 5-14.
- Tabachnick, B., & Fidell, L. (1996). Using multivariate statistics. New York, NY: Harper Collins College Publishers.
- Taylor, S. E., Kemeny, M. E., Bower, J. E., Gruenewald, T. L., & Reed, G. M. (2000). Psychological resources, positive illusions, and health. American Psychologist, 55(1), 99-109.
- Thoresen, C. E. (1999). Spirituality and health: Is there a relationship? Journal of Health Psychology, 4(3), 291-300.
- U.S. Bureau of Census. (1993). Sixty-five plus in America. Current Population Reports (Special Issue, P25-1092). Washington, DC: U.S. Government Printing Office.
- U.S. Department of Health and Human Services, Public Health Service. (1990). Healthy people 2000: National health promotion and disease prevention objectives. Washington, DC: Superintendent of Documents, Government Printing Office.
- Weaver, A. J., Flannelly, L. T., Flannelly, K. J., Koenig, H. G., & Larson, D. B. (1998). An analysis of research and spiritual variables in three major mental health nursing journals, 1991-1995. Issues in Mental Health Nursing, 19, 263-276.

Westgate, C. E. (1996). Spiritual wellness and depression. Journal of Counseling and Development, 75(1), 26-35.

- Witmer, J., & Sweeney, T. (1992). A holistic model for wellness and prevention over the live span. Journal of Counseling and Development, 71, 140-147.
- Witmer, J.M., Rich, C., Barcikowski, R., & Mague, J. C. (1983). Psycosocial characteristics mediating the stress response: An exploratory study. The Personnel and Guidance Journal, 62, 73-77.
- Zinnbauer, B. Pargament, K.I., Cole, B., Rye, M.S., Butter, E.M., Belavich, T.G., et al. (1997). Religion and spirituality: Unfuzzying the fuzzy. *Journal for the Scientific Study of Religion, 36*, 549-564.

#### SHARON SACKS

#### **EDUCATION:**

**Ph.D., Counseling Psychology**, August 2002 Andrews University, Berrien Springs, Michigan.

M.S., Counseling and Human Services, May 1994 Indiana University, South Bend, Indiana

B.S., Public and Environmental Affairs, May 1987

Indiana University, Bloomington, Indiana

#### EXPERIENCE:

#### 2001-2002 Northeastern Illinois University

Pre-doctoral internship - responsibilities consisted of individual and couple's counseling. psychological assessment, report writing, didactic presentations, outreach programs, therapy seminars, professional development, and group work.

#### 1997-1998 Nasr Psychiatric Services

Pre-doctoral practicum - duties consisted of individual and family therapy, psychological assessment, report writing, administering, scoring, and interpreting the: WAIS-III, WAIS-R, MMPI-2, MCMI-II, and the Rorchach-Exner 3<sup>rd</sup> edition. Received individual supervision at the University Counseling Center at the University of Notre Dame.

#### 1996-1997 Andrews University Counseling Center

Pre-doctoral practicum - duties consisted of individual, marital, child, adolescent, and family therapy, as well as psychological assessment and report writing. Administered, scored, and interpreted results for the: WAIS-R, WISC-III, WIAT, MMPI-2, T.A.T., C.A.T., Stanford-Binet Fourth Edition, H-T-P, H-F-D, Bender Gestalt Visual Motor Intelligence Test, and the VMI. Received both individual and group supervision at this site.

#### 1995 LaPorte County Youth Service Bureau

Parenting Program Coordinator - duties consisted of counseling and coordinating parenting program. Coordinated and educated volunteers to go into homes of

clients and teach appropriate parenting and life skills. Counseled troubled youths and parents one-on-one at the center and in the home. Responsible for both the Michigan City office and LaPorte office.

#### 1993 Madison Center - Indiana University at South Bend

Practicum student - duties included individual, career, group, and family therapy Administered, scored, and interpreted the Myers-Briggs-Type Indicator, Self-Directed Search, Strong Interest Inventory, H-T-P, and the H-F-D.

#### **RESEARCH/ASSISTANT EXPERIENCE:**

1996, 1997 Graduate assistant

Dr. W.A. Futcher - Introduction to Statistical Analysis in Psychology and Education . Duties included scoring assignments, statistical projects, and exams, as well as instructing lab twice a week.

#### 1994 - 1995 Research assistant

Dr. Vincent J. Peterson - collected, organized and updated new material for text book, *Orientation to Counseling - Third Edition*. Revised and edited the instructor's manual that accompanied the text book.

#### 1994 Masters thesis

Nonverbal Aggression in Children Expressed Through the H-T-P Projective Drawing Test.